

THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF STREET PARKING PLACES) ORDER 2007

JUNE 2007



CHRISTCHURCH BOROUGH COUNCIL

HEAD OF PLANNING & TRANSPORTATION D JOBBINS BSc (Hons) DipT.P., DipE.M., M.R.T.P.I

DOCUMENT CONTROL

<u>Issue</u>	<u>Date</u>	<u>Purpose</u>	<u>By</u>	<u>Check</u>
ISSUE 1	March 2003	New Consolidated Order	RFB	RFB
ISSUE 2	May 2004	Amended Consolidated Order	RFB	RFB
ISSUE 3	JULY 2005	AMENDED CONSOLIDATED ORDER	RFB	RFB
ISSUE 4	JULY 2006	AMENDED CONSOLIDATED ORDER	RFB	RFB
ISSUE 5	JUNE 2007	Amended Consolidated Order	RFB	RFB

DOCUMENT CONTROL NO.

CD COPY

THIS "CONTROLLED" DOCUMENT IS IDENTIFIED BY THE ABOVE UNIQUE DOCUMENT NUMBER STAMPED IN RED. DOCUMENTS NOT CARRYING THESE RED NUMBERS OR THOSE COPIES ON WHICH THE CONTROL NUMBER IS BLACK (I.E. PHOTOCOPIES, ARE UNCONTROLLED AND THE HOLDER WILL NOT RECEIVE AMENDMENTS OR RE-ISSUES.

Street Name	Page Number
Abbots Close	271
Abington Drive	289
Addiscombe Road	188
Airfield Road	218,231,233,246
Airfield Way	231,246
Amble side	178
Anchor Close	238
Argyle Road	220,233
Arthur Lane	189
Arthur Road	189
Arundel Way	273
Asquith Close	197
Avenue Road	170,186
Avon Buildings	191
Avon Buildings Avon Run Close	239
Avon Run Road	239 239,240,252,254
Avon Run Road	239,240,252,254
	215
Bailey Drive	171
Bargates	182,189,191
Bargates Coop Access	191
Bargates Service Road	191
Barrack Road	156,166,168,170,171,180
Barrack Road Beacon Drive	271
Beaconsfield Road	191
Beaulieu Avenue	169
Beaulieu Avenue Beaulieu Road	169 169,170
Belvedere Road	189
Beivedere Road Bingham Road	229
•	173
Bosley Close	
Bosley Way	173
Brabazon Drive	248
Braemar Drive	274,275,287
Bridge Street	194
Brook Way	259
Bronte Avenue	192
Bub Lane	217
Buccaneers Close	216
Buce Hayes Close	277
Bure Close	235,251
Bure Haven Drive	234,236
Bure Homage Gardens	236
Bure Lane	235,239,240,248,258,259
Bure Road	236
Burley Road	215
Burnett Avenue	166
Burnett Road	169
Burnham Road	205
Burton Road	225
The Buttery	196,204

Street Name	Page Number
Castle Avenue	267,269
Cameron Road	196
Campbell Road	212,213
Capesthorne	240
Castle Street	183,195
Caxton Close	245
Chapel Gate	144,146
Chapel Lane	146
Chestnut Way	213
Chewton Common Road	278,287
	232,238
Chichester Way	191,200,201,202,203,224,225,226,241
Christchurch Bypass	
Church Lane Church Street	183 183
Clarendon Road	183
Coastguard Way	223
Copse Way	267
Court Close	216
Cricket Close	223
Croft Road	230
Cross Way	174
	224
Dairy Close	204
De Havilland Way	234,235
Dennistoun Avenue	231
Devon Road	166
Disraeli Road	197
Dorset Road	241
Douglas Avenue	181
Dragoon Way	170
Draper Road	228,229
Ducking Stool Lane	182
Dunlin Close	239
Earlsdon Way	266
East Cliff Way	251,259
Edward Road	227,242
Elizabeth Avenue	170
Elm Avenue	174
Elphinstone Road	278
Elmwood Way	271,280
Essex Avenue	176
Everest Road	225
Fairway Drive	163,181
Falcon Drive Service Rd	238
Fairfield	190,191
Fairmile Road	173,176,177,186,187,188
Falcon Drive	237,238,239
Farm Lane	238

Street Name	
Fitzmaurice Road	-
Forest Rise	
Forest Way	
Fountain Way	-
Freshwater Road	-
Friars Road	-
Fulmar Road	-
Gardner Road	_
Gleadowe Avenue	
Glengarry Way	
Glenville Road	
Gordon Road	-
Grange Road	-
Greenways	
Greywell	
The Grove	
Groveley Road	
Grove Road West	
Harbour Crescent	_
Heather Close	-
Highcliffe Road	
High Street	_
Highview Close	_
Hillside Drive	-
Hinton Wood Avenue	_
Hoburne Lane	
Holme Road	
Hurn Court Lane	
Hurn Road	
Hunt Road	
Hynesbury Road	
,,	
Island View Avenue	
Jumpers Avenue	
Jumpers Road	
Kings Avenue	
Langley Road	
Latch Farm Avenue	
The Lawns	
Ledbury Road	
Lights Close	
Livingstone Road	
Lodge Road	
Lymington Road	
Lyndhurst Road	

Borough of Christchurch – Street Index

_	
	Page Number
	166,168
	262
	262
	182
	251,253
	236,239
	238
	166
	162,180
	251
	284,285
	276,277
	244,247
	275
	154
	165,168,174,175
	216
	168
	221
	284,286
	244,247,258
	182,183
	178
	178,179
	266,267,271,273,274
	244
	278
	153
	172,179
	225,226,227
1	251,259
1	
1	236,239
1	
1	165,166
	168,186
1	
	162,180
1	
	273
	186
1	279
	222,223
	190
	196,204
1	165
	258,259,269,277,278,279
	241,243,249,250,261,262,263
- 1	

JF/MP/CBC (Con Parking) Order 2007

Click on a page number to travel to that page in the docume

Street Name	Page Number
Magdalen Lane	180
Maple Close	271
Marlow Drive	178
Marlpit Drive	284
Marmion Green	230
Martins Hill Lane	207,208,209
Manor Road	189
The Meadway	262
Medina Way	251
The Meridians	164
Merlin Way	237
Merritown Lane	150
Mill Lane	278
Mill Road	187
Millhams Street	182,183
	182
Millhams Street North Moffat Road	204
Monkswell Green	195
Moorcroft Avenue	205
Montagu Road	279,282
Mudeford	223,238,239
Mudeford Green Close	223
Mudeford Lane	219,222,238
Mude Gardens	238
Nea Road	267
Newlands Road	230
Normandy Drive	204
Oak Avenue	155,165,175
Oakleigh Way	271
Orchard Close	180
Orchid Way	202
Park Close	212
Parkside	265
Parley Lane	143,149,150,151,152
Pauntley Road	217,222
Pennant Road	230
Peregrine Road	234,237
Pine Crescent	267,269
Pinewood Close	284
Pittmore Road	205
Poplar Close	278,279
Portfield Road	170,188
Princess Avenue	183,185
Priors Close	259
Purewell	194,196
Purewell Close	215
	204
Purewell Cross Road	204

Page Number
183,185,198 185
216,217
271
237
271
228
167
187
185
283,284,285,288
203,204,203,200
155,157,158,159,160,161
240
269,270,271
204
167,186
247,248
217
0.11
241
209,210,211,212,214,215
216
172
161
176
181,183,185
256
251,259
204
251,259
278,279
251,258,259
237,239
183
267,271
256,266
256,267
216,229,230,231,242,245
229,230,231,232,245
229
166
181,182,183,184
235,251,252
181
276,277
276,277 216,217,220,221

Street Name	
Stony Lane South	
Stour Road	
Stourbank Road	
Stourvale Avenue	
Stour Way	
Stroud Park Avenue	
Stuart Road	
Suffolk Avenue	
Twynham Avenue	
Tresillian Way	
Valiant Max	
Valiant Way	
Vecta Close	
Victoria Road	
Viking Way	
A/	
Warren Avenue	
Waterford Gardens	
Waterford Place	
Waterford Road	
Watermill Road	
Wharncliffe Gardens	
Wharncliffe Road	
Walkford Road	
Whitehall	
Wickfield Avenue	
Wickfield Close	
Wick Lane	
Willow Drive	
Willow Way	
Wilverley Road	
Woodfield Gardens	
Woodhayes Avenue	
Woodland Way	
Woodstock Road	
Wortley Road	
Wren Close	

Borough of Christchurch – Street Index

_	Derough of officiation officer mack
	Page Number
	194,196
	164,180,181,184,188,189
	180,181
	155,165
	158
	229
	278,279
	176
	180,189
	285
	200
-	248
-	251
-	220
	238
	000 000
_	222,223
	280
	280,282
	278,282
	186
	280
	271,280,281,282
	285,288,290
	183,185
	183
	183
	183,185
	184
	184
	242,245,246
	264
	273
	257,260,267
	206
	277,279,280
	239
1	
1	
_	

JF/MP/CBC (Con Parking) Order 2007

BOROUGH OF CHRISTCHURCH

THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007

The Drawings in this document are copyright of Christchurch Borough Council and may only be reproduced or copied in whole or in part with prior written authorisation to the Head of Planning and Transportation on behalf of the copyright holder

BOROUGH OF CHRISTCHURCH

THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007

The Borough Council of Christchurch ("the Council") in exercise of the powers of the Council under Section 35 of the Road Traffic Regulation Act 1984 ("the Act) and Parts III and IV of Schedule 9 of the Act and under Section 43 and Schedules 3 and 6 of the Road Traffic Act 1991 and of all other enabling powers and other consultation with the Chief Officer of Police in accordance with Part III of Schedule 9 of the Act and with the consent of Dorset County Council in accordance with Section 39(3) of the Act hereby make the following Order:-

CONTENTS

PART I

Definitions, Commencement and Revocation

PART II

Regulation of Use of On-Street Parking Places

PART III

Regulation of Use of Off-Street Parking Places

ATTESTATION CLAUSE

SCHEDULES:-

FIRST SCHEDULE	-	NO WAITING AT ANY TIME
SECOND SCHEDULE	-	NO WAITING AT ANY TIME 01 APRIL TO 30 SEPTEMBER
THIRD SCHEDULE	-	NO WAITING AT ANY TIME 01 MAY TO 30 SEPTEMBER

FOURTH SCHEDULE - NO WAITING AT ANY TIME 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM

- F FIFTH SCHEDULE NO WAITING AT ANY TIME 01 APRIL T0 30 SEPTEMBER 10.00 PM TO 7.00 AM
- SIXTH SCHEDULE NO WAITING 10.00 AM TO 6.00 PM
- SEVENTH SCHEDULE NO WAITING 10.00 AM TO 7.00 PM
- EIGHTH SCHEDULE NO WAITING 8.00 AM TO 6.00 PM
- NINTH SCHEDULE NO WAITING 9.00 AM TO 5.00 PM

ELEVENTH SCHEDULE

THIRTEENTH SCHEDULE

- TENTH SCHEDULE NO WAITING 9.00 AM TO 6.00 PM
 - NO WAITING 9.00 AM TO 7.00 PM
- TWELFTH SCHEDULE NO WAITING MONDAY TO FRIDAY 10.00 AM TO 4.00 PM
 - NO WAITING MONDAY TO FRIDAY 10.00 AM TO 6.00 PM
- FOURTEENTH SCHEDULE NO WAITING MONDAY TO FRIDAY 8.30 AM TO 4.30 PM
- FIFTEENTH SCHEDULE NO WAITING MONDAY TO FRIDAY 8.00 AM TO 6.00 PM
- SIXTEENTH SCHEDULE NO WAITING MONDAY TO FRIDAY 9.00 AM TO 4.00 PM
- SEVENTEENTH SCHEDULE NO WAITING MONDAY TO FRIDAY 8.30 AM TO 9.30 AM AND 3.00PM TO 4.00 PM
- EIGHTEENTH SCHEDULE NO WAITING MONDAY TO FRIDAY 10.00 AM TO 12.00 (NOON)
- NINETEENTH SCHEDULE NO WAITING MONDAY TO SATURDAY 10.00 AM TO 6.00 PM
- TWENTIETH SCHEDULE WAITING LIMITED DISABLED ONLY 10.00 AM TO 4.00 PM 3 HOURS NO RETURN WITHIN ONE HOUR
- TWENTY-FIRST SCHEDULE WAITING LIMITED TO 10.00 AM TO 6.00 PM TWO HOURS NO RETURN WITHIN TWO HOURS

JF/MP/CBC (Con Parking) Order 2007

- TWENTY-SECOND SCHEDULE WAITING LIMITED TO 10.00 AM TO 6.00PM TO TWO HOURS NO RETURN WITHIN 30 MINUTES
- TWENTY-THIRD SCHEDULE WAITING LIMITED TO 10.00 AM TO 6.00 PM TO 30 MINUTES NO RETURN WITHIN 30 MINUTES
- TWENTY-FOURTH SCHEDULE WAITING LIMITED TO 10.00 AM TO 6.00 PM FOR 30 MINUTES NO RETURN WITHIN ONE HOUR
- TWENTY-FIFTH SCHEDULE WAITING LIMITED TO 10.00 AM TO 6.00 PM FOR 30 MINUTES IN ANY HOUR
- TWENTY-SIXTH SCHEDULE WAITING LIMITED TO 10.00 AM TO 6.00 PM FOR ONE HOUR
- TWENTY-SEVENTH SCHEDULE WAITING LIMTED TO 10.00 AM TO 7.00 PM FOR ONE HOUR NO RETURN WITHIN ONE HOUR
- TWENTY-EIGHTH SCHEDULE WAITING LIMITED TO 10.00 AM TO 7.00 PM FOR 30 MINUTES NO RETURN WITHIN 30 MINUTES
- TWENTY-NINETH SCHEDULE WAITING LIMITED TO TWO HOURS NO RETURN WITHIN ONE HOUR
- THIRTIETH SCHEDULE WAITING LIMITED TO TWENTY-FOUR HOUR LIMIT
- THIRTY-FIRST SCHEDULE WAITING LIMITED TO TWENTY-FOUR HOURS NO RETURN WITHIN ONE HOUR NO SLEEPING BETWEEN HOURS OF 9.00 PM TO 6.00 AM
- THIRTY-SECOND SCHEDULE WAITING LIMITED TO 30 MINUTES IN ANY ONE HOUR
- THIRTY-THIRD SCHEDULE WAITING LIMITED TO 30 MINUTES
- THIRTY-FOURTH SCHEDULE WAITING LIMITED TO 60 MINUTES NO RETURN WITHIN 60 MINUTES
- THIRTY-FIFTH SCHEDULE WAITING LIMITED TO 8.00 AM TO 6.00 PM FOR 30 MINUTES NO RETURN WITHIN 30 MINUTES

THIRTY-SIXTH SCHEDULE - PERMIT PARKING ONLY

THIRTY-SEVENTH SCHEDULE - DISABLED PARKING ONLY

- THIRTY-EIGHTH SCHEDULE WAITING LIMITED TO 9.00 AM TO 6.00 PM 30 MINUTES NO RETURN WITHIN 30 MINUTES
- THIRTY-NINETH SCHEDULE WAITING LIMITED 01 APRIL TO 30 SEPTEMBER 24 HOURS NO RETURN WITHIN ONE HOUR
- FORTIETH SCHEDULE WAITING LIMITED 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM ONE HOUR
- FORTY-FIRST SCHEDULE WAITING LIMITED 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM 60 MINUTES NO RETURN WITHIN 30 MINUTES
- FORTY-SECOND SCHEDULE WAITING LIMITED MONDAY TO FRIDAY 10.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR
- FORTY-THIRD SCHEDULEWAITING LIMITED MONDAY TO FRIDAY10.00 AM TO 4.00 PMTWO HOURS NORETURN WITHIN ONE HOUR
- FORTY-FOURTH SCHEDULE WAITING LIMITED MONDAY TO FRIDAY 9.00 AM TO 5.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR (EXCEPT PERMIT HOLDERS)
- FORTY-FIFTH SCHEDULE WAITING LIMITED MONDAY TO FRIDAY 8.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR (EXCEPT PERMIT HOLDERS)
- FORTY-SIXTH SCHEDULE WAITING LIMITED MAY TO SEPTEMBER 10.00 AM TO 6.00 PM 30 MINUTES NO RETURN WITHIN ONE HOUR
- FORTY-SEVENTH SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 10.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR
- FORTY-EIGHTH SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 8.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR

- FORTY-NINETH SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 10.00 AM TO 6.00 PM 30 MINUTES NO RETURN WITHIN 30 MINUTES
- FIFTIETH SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 8.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR (EXCEPT PERMIT HOLDERS)
- FIFTY-FIRST SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 8.00 AM TO 6.00 PM TWO HOURS NO RETURN WITHIN TWO HOURS
- FIFTY-SECOND SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 9.00 AM TO 5.00 PM SIXTY MINUTES NO RETURN WITHIN SIXTY MINUTES (EXCEPT PERMIT HOLDERS)
- FIFTY-THIRD SCHEDULE WAITING LIMITED MONDAY TO SATURDAY 9.00 AM TO 6.00 PM SIXTY MINUTES NO RETURN WITHIN SIXTY MINUTES
- FIFTY-FOURTH SCHEDULE CLEARWAY (NO STOPPING)
- FIFTY-FIFTH SCHEDULE COACHES ONLY WAITING LIMITED FIFTEEN MINUTES NO RETURN WITHIN THIRTY MINUTES
- FIFTY-SIXTH SCHEDULE NO LOADING AT ANY TIME
- FIFTY- SEVENTH SCHEDULE NO LOADING 10.00 AM TO 3.00 PM
- FIFTY- EIGHTH SCHEDULE FIFTY- NINETH SCHEDULE
- SIXTIETH SCHEDULE
- SIXTY-FIRST SCHEDULE
- SIXTY-SECOND SCHEDULE

- NO LOADING 8.00 AM TO 5.00 PM
- NO LOADING AT ANY TIME EXCEPT TO NUMBER 2 SOPERS LANE
- NO LOADING AT ANY TIME EXCEPT TO PREMISES IN CHURCH STREET
- NO LOADING AT ANY TIME EXCEPT TO NUMBERS 19, 21, 23, 25, 30 AND 32 HIGH STREET
- NO LOADING 8.00 AM TO 6.00 PM EXCEPT BY COMMERCIAL VEHICLES TO 45 – 67 ODDS AND 46 – 76 EVENS HIGH
- JF/MP/CBC (Con Parking) Order 2007

STREET

SIXTY-THIRD SCHEDULE - NO LOADING AT ANY TIME EXCEPT TO CHURCH FOR WEDDINGS AND FUNERALS

SIXTY-FOURTH SCHEDULE - NO LOADING AT ANY TIME EXCEPT TO 26 AND 27 MILLHAMS STREET

SIXTY-FIFTH SCHEDULE - NO LOADING 1 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM

- SIXTY-SIXTH SCHEDULE GOODS VEHICLES LOADING ONLY MONDAY TO SATURDAY 8.00 AM TO 6.00 PM 20 MINUTE LIMIT
- SIXTY-SEVENTH SCHEDULE BUS STOP CLEARWAY 7.00 AM TO 7.00 PM (NO STOPPING)
- SIXTY-EIGHTH SCHEDULE HACKNEY CARRIAGE STAND (TAXIS) (NO STOPPING)
- SIXTY-NINETH SCHEDULE NO STOPPING MONDAY TO FRIDAY 8.00 AM TO 5.00 PM ON SCHOOL ENTRANCE MARKINGS

SEVENTIETH SCHEDULE - PARKING CHARGES FOR OFF-STREET PARKING PLACES

ANNEXURES

On-Street Plans

Off-Street Plans

PART I

DEFINITIONS, COMMENCEMENT AND REVOCATION

1.0 **DEFINITIONS**

- 1.1 In Parts II and III of this Order except where the context otherwise requires the following expressions shall have the following meanings:
 - 1.1.1 "**Approved**" means approved by the Secretary of State for the Environment
 - 1.1.2 **"Authorised Cab Rank"** means an area of carriageway which is comprised within and indicated by a road marking complying with diagram 1028.2 in Schedule 6 of the Traffic Signs Regulations and General Directions 2002
 - 1.1.3 "Authorised Vehicle" means a vehicle specified in Column (3) of Part I of the Seventieth Schedule and complying with any height restriction imposed by the Council and shall include electrically propelled vehicles intended or adapted for use on roads
 - 1.1.4 "Bus Stop" means an area on a highway which is bounded by a marking shown on diagram 1025 in Schedule 6 of the Traffic Signs Regulations and General Directions 2002
 - 1.1.5 "**Caravan**" is as defined in Section 29 of the Caravan Sites and Control of Development Act 1960
 - 1.1.6 **"Coach"** means a vehicle adapted to carrying more than 18 passengers exclusive of the driver
 - 1.1.7 **"Collecting**" and **"Delivering"** in relation to any goods includes checking the goods for the purpose of their collection or delivery
 - 1.1.8 "Disabled Driver" means (i) any person driving an invalid carriage or (ii) any person driving an unauthorised vehicle displaying a badge issued in accordance with the Disabled Persons (Badges for Motor Vehicles) (England) Regulations 2000 as amended

- 1.1.9 "Disabled Person" means a disabled person of a description prescribed by Regulation 4 of the Disabled Persons (Badges for Motor Vehicles) (England) Regulations 2000 as amended
- 1.1.10 "Disabled Persons Badge" has the same meaning as in Regulation 11 of the Disabled Persons (Badges for Motor Vehicles) (England) Regulations 2000 as amended
- 1.1.11 "**Disabled Persons Parking Disc**" should be displayed as described in the Local Authorities' Traffic Orders (Exemptions for Disabled Persons) (England) Regulations 2000 as amended
- 1.1.12 "Disabled Persons Vehicle" has the same meaning as in the Local Authorities' Traffic Orders (Exemptions for Disabled Persons) (England) Regulations 2000 as amended
- 1.1.13 "**Driver**" in relation to a vehicle waiting in a parking place means the person driving the said vehicle or responsible for the vehicle at the time it was left in the parking place. The registered keeper of the said vehicle shall be deemed to have been the driver unless evidence to the contrary is shown
- 1.1.14 "**Goods Vehicle**" means a vehicle constructed or adapted for use for the conveyance of goods or burden the gross weight of which is between 3500kg and 7.5 tonnes
- 1.1.15 "**Hackney Carriage**" has the same meaning as in Section 38 of the Town Police Clauses Act 1847
- 1.1.16 "Hackney Carriage Stand" means any area of the carriageway of a specified road intended for the waiting of hackney carriages which is comprised within and indicated by a road marking complying with diagram 1028.2 in Schedule 6 of the Traffic Signs Regulations and General Directions 2002
- 1.1.17 "Heavy Commercial Vehicle" is as defined in Section 138 of the Act
- 1.1.18 "Heavy Motor Car" is as defined in Section 136(3) of the Act
- 1.1.19 "Invalid Carriage" is as defined by Section 136(5) of the Act
- 1.1.20 "Motor Car" is as defined in Section 136(2)(a) of the Act

- 1.1.21 "**No Loading Orders**" means any Orders made by the Council or its predecessors in relation to restrictions on loading and unloading on roads within the administrative area of the Borough of Christchurch
- 1.1.22 "**No Waiting Orders**" means any Orders made by the Council or its predecessors in relation to restrictions on waiting on roads within the administrative area of the Borough of Christchurch
- 1.1.23 "Notice to Owner" means the form produced by the Council headed "Notice to Owner/Keeper/Hirer"
- 1.1.24 "**Off-Street Plans**" means the plans annexed to this Order headed "Off-Street" Plans
- 1.1.25 **"On-Street Plans"** means the plans annexed to this Order headed 'On-Street Plans'
- 1.1.26 "Owner" means the person by whom the vehicle is kept and used and in determining for the purposes of this Order who was the owner of the vehicle at the time it shall be presumed that the owner was the person in whose name the vehicle was at that time registered under the Vehicle Excise and Registration Act 1994 and where the vehicle is the subject of a hiring agreement or hire purchase agreement "Owner" includes the person entitled to possession of the vehicle under the agreement
- 1.1.27 "**Owner of Business Premises**" means a person who is the sole proprietor or partner of business premises situated in the Borough of Christchurch but excluding Company Shareholders
- 1.1.28 "**Parking Attendant**" means any person duly authorised by or on behalf of the Council or Police Constable in uniform to supervise the Parking Place
- 1.1.29 "**Parking Bay**" means the part of the Parking Place marked out for the leaving of an Authorised Vehicle
- 1.1.30 "Parking Disc" means a disc issued by a local authority complying with the requirements of the British Standard Specification for Parking Discs (BS No 4631:70), and capable of showing the quarter hour period during which a period of waiting begins

- 1.1.31 "Parking Place" means any area of land specified by name in Column (1) of the Seventieth Schedule and more particularly shown on the Off-Street Plans hereinafter defined for use as a Parking Place
- 1.1.32 "**Passenger Vehicle**" means a vehicle adapted to carry more than 8 passengers exclusive of the driver
- 1.1.33 "**Public Service Vehicle**" has the same meaning as in Section 1 of the Public Passenger Vehicles Act 1981
- 1.1.34 "**Public Service Vehicle Orders**" means any Orders made by the Council or its predecessors in relation to restrictions on roads within the administrative area of the Borough of Christchurch relating to public service vehicles, hackney carriage vehicles and taxis
- 1.1.35 "**Resident of the Borough**" means a person who is resident within the Borough of Christchurch and whose name would normally appear on the Electoral Register prepared for the said Borough
- 1.1.36 **"Season Ticket**" and "**Parking Permit**" means a ticket or permit issued in accordance with the provisions of Part III of this Order
- 1.1.37 "Taxi" means a licensed taxi as described in Section 37 of the Town Police Clauses Act 1847 and licensed hire car as licensed under Section 48 of the Local Government (Miscellaneous Provisions) Act 1976
- 1.1.38 "**Ticket Machine**" means an apparatus of an approved type and design being an apparatus designed to issue parking tickets indicating the payment of a charge and the day and time at which the charge was paid
- 1.1.39 "**Trailer**" includes where the context allows a Caravan and for the purposes of this Order the Trailer and the Vehicle by which it is conveyed shall be deemed to be separate Vehicles and the Driver of the Vehicle shall be deemed to be the Driver of the Trailer

1.2 Further Provisions relating to Interpretation relating to On-Street and Off-Street Parking

- 1.2.1 For the purposes of this Order a vehicle shall be regarded as displaying a Disabled Person's Badge in the relevant position when and only when:-
 - (i) in the case of a vehicle fitted with a dashboard or facia panel the badge is exhibited thereon so that Part I of the badge is legible from the outside of the vehicle; or
 - (ii) in the case of a vehicle fitted with a dashboard or facia panel the parking disc is exhibited thereon so that the quarter hour period during which the period of waiting began is legible from the outside of the vehicle;
 - (iii) in the case of a vehicle not so fitted the parking disc is exhibited in a conspicuous position on the vehicle so that the quarter hour period during which the period of waiting began is legible from outside the vehicle
- 1.2.2 For the purposes of this Order a vehicle shall be deemed to wait if for more than two minutes in the same place any one part of the road is below any part of the vehicle or its load (if any) throughout the period exceeding two minutes whether or not the vehicle is moved during the period
- 1.2.3 For the avoidance of doubt reference to a "Vehicle" in this Order includes a Coach, Goods Vehicle, Hackney Carriage, Heavy Commercial Vehicle, Heavy Motor Car, Invalid Carriage, Motor Car, Passenger Vehicle, Public Service Vehicle, Taxi or a Trailer as appropriate
- 1.3 Interpretation Act 1978 shall apply for the interpretation of the Order as it applies for the interpretation of an Act of Parliament

2.0 COMMENCEMENT

2.1 This Order shall come into operation on the day of of the year Two Thousand and Six 2.2 The restrictions imposed by this Order are in addition to and not in derogation of any restrictions or requirements imposed by any regulations made or having effect as if made under the Act or any other enactment.

3.0 REVOCATION

- 3.1 The Borough of Christchurch (Various Roads and Parking Places) (Consolidation of Orders made for On and Off-Street Parking Places) Order 2005 is revoked from the Commencement Date
- 3.2 All previous variation and amendment orders made by the Council in respect of on and off-street parking places specified in the Schedules to this Order are hereby revoked from the Commencement Date

PART II

REGULATION OF USE OF ON-STREET PARKING PLACES

Article No	Description
1.0	No Waiting Restrictions
2.0	Loading Restrictions
3.0	Public Service Vehicles
4.0	Provisions relating to On-Street Parking
5.0	Liability to and Amount of Penalty
	Charge

1.0 WAITING RESTRICTIONS

1.1 No Waiting at Any Time

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of the roads specified in the First Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.2 No Waiting 01 April to 30 September at Any Time

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time from 01 April to 30 September inclusive in any year on the roads or sides of the roads specified in the Second Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.3 No Waiting 01 May to 30 September at Any Time

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time from 01 May to 30 September inclusive in any year on the roads or sides of the roads specified in the Third Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.4 No Waiting 01 April to 30 September 10.00 am to 7.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time from 01 April to 30 September inclusive in any year between the hours of 10.00 am to 7.00 pm on the roads or sides of the roads specified in the Fourth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.5 No Waiting 01 April to 30 September 10.00 pm to 7.00 am

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time from 01 April to 30 September inclusive in any year between the hours of 10.00 pm to 7.00 am on the roads or sides of the roads specified in the Fifth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.6 No Waiting 10.00 am to 6.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 6.00 pm in any year on the roads or sides of the roads specified in the Sixth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.7 No Waiting 10.00 am to 7.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 7.00 pm on the roads or sides of the roads specified in the Seventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.8 **No Waiting 8.00 am to 6.00 pm**

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 8.00 am to 6.00 pm on the roads or sides of the roads specified in the Eighth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.9 **No Waiting 9.00 am to 5.00 pm**

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 9.00 am to 5.00 pm on the roads or sides of the roads specified in the Ninth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.10 No Waiting 9.00 am to 6.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 9.00 am to 6.00 pm on the roads or sides of the roads specified in the Tenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.11 No Waiting 9.00 am to 7.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 9.00 am to 7.00 pm on the roads or sides of the roads specified in the Eleventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.12 No Waiting Monday to Friday 10.00 am to 4.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 4.00 pm on Monday to Friday on the roads or sides of the roads specified in the Twelfth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.13 No Waiting Monday to Friday 10.00 am to 6.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 6.00 pm on Monday to Friday on the roads or sides of the roads specified in the Thirteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.14 No Waiting Monday to Friday 8.30 am to 4.30 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 8.30 am to 4.30 pm on Monday to Friday on the roads or sides of the roads specified in the Fourteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.15 No Waiting Monday to Friday 8.00 am to 6.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 8.00 am to 6.00 pm on Monday to Friday on the roads or sides of the roads specified in the Fifteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.16 No Waiting Monday to Friday 9.00 am to 4.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 9.00 am to 4.00 pm on Monday to Friday on the roads or sides of the roads specified in the Sixteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.17 No Waiting Monday to Friday 8.30 am to 9.30 am and 3.00 pm to 4.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 9.30 am to 9.30 am and 3.00 pm to 4.00 pm on Monday to Friday on the roads or sides of the roads specified in the Seventeenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.18 No Waiting Monday to Friday 10.00 am to 12.00 (noon)

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 12.00 noon on Monday to Friday on the roads or sides of the roads specified in the Eighteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.19 No Waiting Monday to Saturday 10.00 am to 6.00 pm

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 6.00 pm on Monday to Saturday on the roads or sides of the roads specified in the Nineteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.20 Waiting Limited Disabled Only 10.00 am to 4.00 pm for Three Hours No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to between the hours of 10.00 am to 4.00 pm on the roads or sides of the roads specified in the Sixteenth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order other than a Disabled Person displaying a valid Disabled Person's Badge and Disabled Person's Parking Disc which Disabled Person shall

be permitted to wait for a period not exceeding three hours between the hours of 10.00am to 4.00pm with no return within one hour on the said roads or sides of the said roads specified in the Twentieth Schedule to this Order

1.21 Waiting Limited to 10.00 am to 6.00 pm for Two Hours No Return within Two Hours

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than two hours between the hours of 10.00 am to 6.00 pm with no return within two hours on the roads or sides of the roads specified in the Twenty-First Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.22 Waiting Limited to 10.00 am to 6.00 pm for Two Hours No Return within Thirty Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than two hours between the hours of 10.00 am to 6.00 pm with no return within thirty minutes on the roads or sides of the roads specified in the Twenty-Second Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.23 Waiting Limited to 10.00 am to 6.00 pm for 30 Minutes No Return within 30 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes between the hours of 10.00 am to 6.00 pm with no return within 30 minutes on the roads or sides of the roads specified in the Twenty-Third Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.24 Waiting Limited to 10.00 am to 6.00 pm for 30 Minutes No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes between the hours of 10.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Twenty-Fourth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.25 Waiting Limited to 10.00 am to 6.00 pm for 30 Minutes in Any Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes in any hour between the hours of 10.00 am to 6.00 pm on the roads or sides of the roads specified in the Twenty-Fifth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.26 Waiting Limited to 10.00 am to 6.00 pm for One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour between the hours of 10.00 am to 6.00 pm on the roads or sides of the roads specified in the Twenty-Sixth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.27 Waiting Limited to 10.00 am to 7.00 pm for One Hour No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour between the hours of 10.00 am to 7.00 pm with no return within one hour on the roads or sides of the roads specified in the Twenty-Seventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.28 Waiting Limited to 10.00 am to 7.00 pm for 30 Minutes No Return within 30 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes between the hours of 10.00 am to 7.00 pm with no return within 30 minutes on the roads or sides of the roads specified in the Twenty-Eighth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.29 Waiting Limited to Two Hours

No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than two hours with no return within one hour on the roads or sides of the roads specified in the Twenty-Ninth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.30 Waiting Limited to Twenty-Four Hour Limit

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than twenty-four hours on the roads or sides of the roads specified in the Thirtieth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.31 Waiting Limited to Twenty-Four Hours No Return within One Hour No Sleeping 9.00 pm to 6.00 am

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than twenty-four hours with no return within one hour and no sleeping in any vehicle between the hours of 9.00 pm and 6.00 am on the roads or sides of the roads specified

in the Thirty-First Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.32 Waiting Limited to 30 Minutes in Any One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes in any one hour on the roads or sides of the roads specified in the Thirty-Second Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.33 Waiting Limited to 30 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes on the roads or sides of the roads specified in the Thirty-Third Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.34 Waiting Limited to 60 Minutes

No Return within 60 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 60 minutes with no return within 60 minutes on the roads or sides of the roads specified in the Thirty-Fourth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.35 Waiting Limited to 8.00 am to 6.00 pm for 30 Minutes No Return within 30 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes between the hours of 8.00 am to 6.00 pm with no return within 30 minutes on the

roads or sides of the roads specified in the Thirty-Fifth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.36 Permit Parking Only

Save as provided in Articles 4.1 and 4.5 to this Order no person shall except upon direction or with the permission of a Parking Attendant cause or permit any vehicle to wait in the Parking bays on the roads or sides of the roads specified in the Thirty-Sixth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.37 Disabled Parking Only

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction of a Parking Attendant cause or permit any vehicle at any time on the roads or sides of the roads specified in the Thirty-Seventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order other than a Disabled Person displaying a valid Disabled Person's Badge and Disabled Person's Parking Disc which Disabled Person shall be permitted to wait

1.38 Waiting Limited to 9.00 am to 6.00 pm 30 Minutes No Return within 30 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes between the hours of 9.00 am to 6.00 pm with no return within 30 minutes on the roads or sides of the roads specified in the Thirty-Eighth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.39 Waiting Limited 01 April to 30 September Twenty-Four Hours No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than twenty-four hours from 01 April to 30 September inclusive in any year with no return within one

hour on the roads or sides of the roads specified in the Thirty-Ninth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.40 Waiting Limited 01 April to 30 September 10.00 am to 7.00 pm One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour from 01 April to 30 September inclusive in any year between the hours of 10.00 am to 7.00 pm on the roads or sides of the roads specified in the Fortieth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.41 Waiting Limited 01 April to 30 September 10.00 am to 7.00 pm 60 Minutes

No Return within 30 Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 60 minutes from 01 April to 30 September inclusive in any year between the hours of 10.00 am to 7.00 pm with no return within 30 minutes on the roads or sides of the roads specified in the Forty-First Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.42 Waiting Limited Monday to Friday 10.00 am to 6.00 pm One Hour No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour on Monday to Friday between the hours of 10.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Second Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.43 Waiting Limited Monday to Friday 10.00 am to 4.00 pm Two Hours No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than two hours on Monday to Friday between the hours of 10.00 am to 4.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Third Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.44 Waiting Limited Monday to Friday 9.00 am to 5.00 pm One Hour No Return within One Hour (except Permit Holders)

Save as provided in Articles 4.1, 4.2, 4.3 and 4.6 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour on Monday to Friday between the hours of 9.00 am to 5.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Fourth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.45 Waiting Limited Monday to Friday 8.00 am to 6.00pm One Hour No Return within One Hour (except Permit Holders)

Save as provided in Articles 4.1, 4.2, 4.3 and 4.6 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour on Monday to Friday between the hours of 8.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Fifth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.46 Waiting Limited May to September 10.00 am to 6.00 pm 30 Minutes No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than 30 minutes between May to September inclusive in any year between the hours of 10.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Sixth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.47 Waiting Limited Monday to Saturday 10.00 am to 6.00 pm One Hour No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour on Monday to Saturday between the hours of 10.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Seventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.48 Waiting Limited Monday to Saturday 8.00 am to 6.00 pm One Hour No Return within One Hour

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour on Monday to Saturday between the hours of 8.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Forty-Eighth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.49 Waiting Limited Monday to Saturday 10.00 am to 6.00 pm Thirty Minutes

No Return within Thirty Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than thirty minutes on Monday to Friday between the hours of 10.00 am to 6.00 pm with no return within thirty minutes on the roads or sides of the roads specified in the Forty-Ninth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.50 Waiting Limited Monday to Saturday 8.00 am to 6.00 pm One Hour No Return within One Hour (except Permit Holders)

Save as provided in Articles 4.1, 4.2, 4.3 and 4.7 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than one hour on Monday to Saturday between the hours of 8.00 am to 6.00 pm with no return within one hour on the roads or sides of the roads specified in the Fiftieth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.51 Waiting Limited Monday to Saturday 8.00 am to 6.00 pm Two Hours No Return within Two Hours

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than two hours on Monday to Saturday between the hours of 8.00 am to 6.00 pm with no return within two hours on the roads or sides of the roads specified in the Fifty-First Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.52 Waiting Limited Monday to Saturday 9.00 am to 5.00 pm Sixty Minutes

No Return within Sixty Minutes (except Permit Holders)

Save as provided in Articles 4.1, 4.2, 4.3 and 4.8 of this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than sixty minutes on Monday to Saturday between the hours of 9.00 am to 5.00 pm with no return within sixty minutes on the roads or sides of the roads specified in the Fifty-Second Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.53 Waiting Limited Monday to Saturday 9.00 am to 6.00 pm Sixty Minutes

No Return within Sixty Minutes

Save as provided in Articles 4.1, 4.2 and 4.3 of this Order no person shall except upon direction or with the permission of a Parking Attendant cause or permit any vehicle to wait for longer than sixty minutes on a Monday to Saturday between the hours of 9.00 am to 6.00 pm with no return within sixty minutes on the roads or sides of the roads specified in the Fifty-Third Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.54 Clearway (No Stopping)

Save as provided in Article 4.1 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of the roads specified in the Fifty-Fourth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

1.55 Coaches Only Waiting Limited Fifteen Minutes No Return within 30 Minutes

Save as provided in Articles 4.1, 4.2, 4.3 and 4.4 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or

sides of the roads specified in the Fifty-Fifth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

2.0 LOADING RESTRICTIONS

2.1 No Loading at Any Time

No person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of the roads specified in the Fifty-Sixth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.2 No Loading 10.00 am to 3.00 pm

No person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 10.00 am to 3.00 pm on the roads or sides of the roads specified in the Fifty-Seventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.3 No Loading 8.00 am to 5.00 pm

No person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 8.00 am to 5.00 pm on the roads or sides of the roads specified in the Fifty-Eighth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.4 No Loading at Any Time Except to Number 2 Sopers Lane

Save as provided in Article 4.9 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of the roads specified in the Fifty-Ninth Schedule to this Order in the areas identified on the OnStreet Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.5 No Loading at Any Time Except to Premises in Church Street

Save as provided in Article 4.10 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of the roads specified in the Sixtieth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.6 No Loading at Any Time Except to Numbers 19, 21, 23, 25, 30 and 32 High Street

Save as provided in Article 4.11 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of the roads specified in the Sixty-First Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.7 No Loading 8.00 am to 6.00 pm Except by Commercial Vehicles to 45 – 67 Odds, and 46 – 76 Evens, High Street

Save as provided in Article 4.12 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 8.00 am to 6.00 pm on the roads or sides of the roads specified in the Sixty-Second Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.8 No Loading at Any Time Except to Church for Weddings and Funerals

No person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time except during wedding and funeral services on the roads or sides of roads specified in the Sixty-Third Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.9 No Loading at Any Time Except to Numbers 26 and 27 Millhams Street

No person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time on the roads or sides of roads specified in the Sixty-Fourth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.10 No Loading 01 April to 30 September 10.00 am to 7.00 pm

No person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time from 01 April to 30 September between the hours of 10.00 am to 7.00 pm on the roads or sides of the roads specified in the Sixty-Fifth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order for the purpose of enabling goods to be loaded on to or unloaded from the vehicle

2.11 Goods Vehicles Loading Only Monday to Saturday 8.00 am to 6.00pm

20 Minutes Limit

Save as provided in Article 4.15 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait Monday to Saturday between the hours of 8.00 am to 6.00 pm on the roads or sides of the roads specified in the Sixty-Sixth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order, **except for goods vehicles for no longer than 20 minutes**, for the purpose of enabling goods to be loaded on to or unloaded from the vehicle
2.12 No Stopping Monday to Friday 8.00am to 5.00pm on School Entrance Markings (Zig-Zags)

Save as provided in Article 4.1, 4.2 and 4.3 of this Order no person shall except upon direction or with the permission of a Parking Attendant cause or permit any vehicle to wait Monday to Friday between the hours of 8.00 am to 5.00 pm on the areas of carriageway of the roads specified in the Sixty-Ninth Schedule to this Order, being areas at or near the access to a school marked by the traffic signs specified in diagrams 642.2A and 1027.1 in the Traffic Signs Regulations and General Directions 2002.

3.0 PUBLIC SERVICE VEHICLES

3.1 Bus Stop Clearways 7.00 am to 7.00 pm (No Stopping)

Save as provided in Article 4.1 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait at any time between the hours of 7.00 am to 7.00 pm in the Bus Stop Clearways on the roads or sides of the roads specified in the Sixty-Seventh Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

3.2 Hackney Carriage Stand (Taxis) (No Stopping)

Save as provided in Articles 4.1 and 4.14 to this Order no person shall except upon the direction or with the permission of a Parking Attendant cause or permit any vehicle to wait on the roads or sides of the roads specified in the Sixty-Eighth Schedule to this Order in the areas identified on the On-Street Plans annexed to this Order

4.0 PROVISIONS RELATING TO ON-STREET PARKING

4.1 Nothing in Articles 1.0, 2.0 and 3.0 shall render it unlawful to cause or permit any vehicle to wait in any areas of roads specified therein for so long as may be necessary to enable:-

4.1.1 a person to board or alight from the vehicle

- 4.1.2 the vehicles, if it cannot conveniently be used for such purposes in any other road, to be used in connection with any of the following operations namely:-
 - (i) building, industrial or demolition operations
 - (ii) the removal of any obstruction to traffic
 - (iii) the maintenance, improvement or re-construction of any at the said areas of road specified; or
 - (iv) the laying, erection, alteration or repair in, or land adjacent to any of the said areas of road specified at any sewer or of any main, pipe or apparatus for the supply of gas, water or electricity, or of any telegraphic lines as defined in the Telecommunications Code contained in Schedule 2 of the Telecommunications Act 1984
- 4.1.3 the vehicle, if it cannot conveniently be used for such purpose in any other road, to be used in the service of a local authority or a water authority in pursuance of statutory powers or duties;
- 4.1.4 the vehicle to be used for the purpose of delivering or collecting postal packets as defined in Section 125(1) of the Postal Services Act 2000;
- 4.1.5 the vehicle to take in petrol, oil, water or air from any garage situated in or adjacent to any of the said areas of roads specified
- 4.1.6 the vehicle to wait at or near to any premises situated in or adjacent to any of the said areas of roads specified for so long as such waiting by that vehicle is reasonably necessary in connection with any wedding or funeral; or
- 4.1.7 the vehicle to be used for fire brigade, ambulance or police purposes

- 4.2 Nothing in Articles 1.0, 2.0 and 3.0 shall render it unlawful to cause or permit any vehicle to wait for so long as may be necessary for the purpose of enabling goods to be loaded on or unloaded from the vehicle in any of the areas of roads specified in the schedules to this Order at any time on any day
- 4.3 Nothing in Articles 1.0, 2.0 and 3.0 shall render it unlawful to cause or permit a disabled person's vehicle which displays in the relevant position a disabled person's badge and a parking disc (on which the driver, or other person in charge of the vehicle, has marked the time at the period of waiting began) to wait in any of the areas of roads specified for a period not exceeding three hours (not being a period separated by an interval of less than one hour from a previous period of waiting by the same vehicle in the same area of road specified on the same day)
- 4.4 Nothing in Article 1.52 shall render it unlawful to cause or permit a Coach to wait in the roads specified in the Fifty-Fifth Schedule to this Order in the areas marked on the On-Street Plans for no longer than fifteen minutes with no return within thirty minutes only
- 4.5 Nothing in Article 1.33 shall render it unlawful for a resident of Ducking Stool Lane or Millhams Street and who has a current Ducking Stool Lane and Millhams Street Parking Permit to cause or permit any vehicle to wait in the roads specified in the Thirty-Sixth Schedule to this Order in the areas shown marked on the On-Street Plans
- 4.6 Nothing in Article 1.41 shall render it unlawful for a resident of Grange Road and who has a current Grange Road Parking Permit to cause or permit any vehicle to wait in the roads specified in the Forty-Fourth Schedule to this Order in the areas shown marked on the On-Street Plans
- 4.7 Nothing in Article 1.47 shall render it unlawful for a resident of Avon Buildings, Fairfield, Princess Avenue, Quay Road, Station Road and Twynham Avenue and who holds a current Avon Buildings Parking Permit,

Fairfield Parking Permit, Princess Avenue Parking Permit, Quay Road Parking Permit or Station Road and Twynham Avenue Parking Permit respectively to cause or permit any vehicle to wait in the roads specified in the Fiftieth Schedule to this Order in the areas shown marked on the On-Street Plans

- 4.8 Nothing in Article 1.49 shall render it unlawful for a resident of Beaconsfield Road and who holds a current Beaconsfield Road Parking Permit to cause or permit any vehicle to wait in the road specified in the Fifty-Second Schedule to this Order in the areas shown marked on the On-Street Plans
- 4.9 Nothing in Article 1.42 shall render it unlawful for a resident of Riverlea Road and who holds a current Riverlea Road Parking Permit to cause or permit any vehicle to wait in the road specified in the Forty-Fifth Schedule to this Order in the areas shown marked on the On-Street Plans
- 4.10 Nothing in Article 2.3 shall render it unlawful to cause or permit any vehicle to wait in the roads specified in the Fifty-Ninth Schedule to this Order in the areas shown marked on the On-Street Plans for the purpose of loading or unloading goods to or from a vehicle in relation to Number 2 Sopers Lane only
- 4.11 Nothing in Article 2.4 shall render it unlawful to cause or permit any vehicle to wait in the roads specified in the Sixtieth Schedule to this Order in the areas shown marked on the On-Street Plans for the purpose of loading or unloading goods to or from a vehicle in relation to premises in Church Street only
- 4.12 Nothing in Article 2.5 shall render it unlawful to cause or permit any vehicle to wait in the roads specified in the Sixty-First Schedule to this Order in the areas shown marked on the On-Street Plans for the purpose of loading or unloading goods to or from a vehicle in relation to Numbers 19, 21, 23, 25, 30 and 32 High Street only

- 4.13 Nothing in Article 2.6 shall render it unlawful to cause or permit any vehicle to wait in the roads specified in the Sixty-Second Schedule to this Order in the areas shown marked on the On-Street Plans for the purpose of loading or unloading goods to or from a vehicle in relation to commercial vehicles to 45 67 (inclusive) odds, and 46 76 (inclusive) evens, High Street only between the hours of 8.00 am and 6.00 pm
- 4.14 Nothing in Article 3.2 shall render it unlawful to cause or permit a Hackney Carriage to wait in the roads specified in the Sixty-Eighth Schedule to this Order in the areas marked on the On-Street Plans
- 4.15 Nothing in Article 2.10 shall render it unlawful to cause or permit any 'goods vehicle' to wait in the roads specified in the Sixty-Sixth Schedule to this Order in the areas shown marked on the On-Street Plans for no longer than 20 minutes for the purpose of loading or unloading goods to or from a vehicle in Wick Lane, Monday to Saturday between the hours of 8.00 am and 6.00 pm

5.0 LIABILITY TO AND AMOUNT OF PENALTY CHARGE

- 5.1 Subject to Article 5.2 the Owner of a vehicle shall be liable to pay an amount of Sixty Pounds (£60) ("the Penalty Charge") if the vehicle is caused or permitted to wait at any time in any of the roads specified in Schedule One to Sixty-Ninth inclusive to this Order in the areas shown marked on the On-Street Plans contrary to any of the Articles in part II of this Order
- 5.2 In the event that the Penalty Charge is paid before the end of the period of 14 days beginning with the date of the Penalty Charge Notice, the amount of the Penalty Charge will be reduced to Thirty Pounds (£30)
- 5.3 Where there is a contravention of any of the Articles in Part II of this Order a Parking Attendant shall attach to the vehicle in a conspicuous position a

notice ("the Penalty Charge Notice") and also referred to as a "PCN" which shall include the following particulars:-

- (a) the registration mark of the vehicle or where the vehicle is being used under a trade licence the number of trade plate carried by the vehicle
- (b) the time at which the Parking Attendant observed that the Penalty Charge had been incurred
- the grounds on which the Parking Attendant believes that a Penalty Charge is payable with respect to the vehicle
- (d) a statement that the Penalty Charge of Sixty Pounds (£60) is required to be paid
- (e) that the Penalty Charge must be paid before the end of the period of 28 days beginning with the date of the Penalty Charge Notice
- (f) that if the Penalty Charge is paid before the end of the period of 14 days beginning with the date of the Penalty Charge Notice, the amount of the Penalty Charge will be reduced to Thirty Pounds (£30)
- (g) that if the Penalty Charge is not paid before the end of the 28 day period, a Notice to Owner may be served by the Council on the person appearing to them to be the Owner of the vehicle
- (h) the address to which payment of the Penalty Charge must be sent
- 5.4 The Penalty Charge shall be paid to the Council:-
- (a) by post with payment being made by crossed cheque, bankers draft, money order or postal order made payable to 'Christchurch Borough Council' which shall be delivered to or sent by post to the Council's Resources Department at the Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ quoting the Penalty Charge Notice number;
- (b) in person by attending the Council Offices at the address above during office hours being 9.00am to 4.30pm Monday to Friday inclusive (excluding Bank Holidays) with payment being made by cash, crossed

cheque, bankers draft, money order or postal order made to 'Christchurch Borough Council' or by credit or debit card;

- (c) by telephone call to the Council Offices on telephone number (01202)
 495252 with payment being made by credit or debit card;
- (d) on-line via the internet by visiting <u>www.dorsetforyou.com</u> and following the on-line instructions with payment being made by credit or debit card

Provided that if the Twenty-Eight day period expires upon a day on which the said Department is closed the period within which payment of the Penalty Charge shall be paid to the Council shall be extended until 1pm on the next full day on which the said Department is open

- 5.5 The period for which a vehicle may be left after the Penalty Charge has been incurred shall not exceed twelve hours and the Driver of the vehicle shall not cause or permit the vehicle to wait for longer than that
- 5.6 If a vehicle is left for more than the period of twelve hours referred to in Article 5.5 after the Penalty Charge has been incurred it shall be the duty of a Parking Attendant to attach to the vehicle in a conspicuous position a further Penalty Charge Notice which shall include the particulars mentioned in Article 5.3 and the following additional particulars namely the time at which he or she first noticed the vehicle had been left for more than the prescribed period after the Penalty Charge had been incurred

PART III

REGULATION OF USE OF OFF-STREET PARKING PLACES

Article No	Description
6.0	Use of the Parking Place
7.0	Council's Powers
8.0	Maximum Period of Waiting
9.0	Charges
10.0	Season Tickets and Parking Permits
11.0	Method of Payment of Charges
12.0	Display of Tickets
13.0	Non-Transfer of Tickets
14.0	Damaged or Foreign Coins
15.0	Liability To and Amount of Penalty Charge
16.0	Maximum Period for which an Authorised Vehicle may be left
	after Penalty Charge incurred
17.0	Restrictions on Removal of Tickets and Penalty Charge
	Notices
18.0	Manner of Payment of Penalty Charge
19.0	Indications by Ticket Machines and Tickets as Evidence
20.0	Exemption from Charges
21.0	Speed of Vehicles
22.0	Restriction on Class of Vehicle using Parking Places and on
	Manner of Parking
23.0	Drivers to Switch off Engine of Authorised Vehicle
24.0	Authorised Vehicles in Parking Places to be in Satisfactory
	Condition
25.0	Driver not to Sound the Horn of Authorised Vehicle in
	Parking Place
26.0	Authorised Vehicle in Parking Place not to be Driven except
	for Entering or Leaving
27.0	Prohibition of Loud Speakers etc
28.0	Prohibition of Breaches of the Peace
	lan 000 7

29.0 Restrictions on the Use of an Authorised Vehicle in	ו the
Parking Place	
30.0 Prohibition of Passage through Car Park	
31.0 Directions for Traffic to be Followed	
32.0 Restrictions on Use of the Parking Place	
33.0 Application of Enactment	
34.0 Removal etc of Vehicles from a Parking Place	
35.0 Manner of Removal of Vehicles from the Parking Place	
36.0 Safe Custody of Vehicles removed from the Parking Place	ce

6.0 USE OF THE PARKING PLACE

- 6.1 Subject to the following provisions of this Order the Parking Place specified by name in Column (1) of Part I of the Seventieth Schedule may be used as a Parking Place for Authorised Vehicles in such positions on such days during such hours and on payment of such charges as are specified in relation to such Parking Place in Part I of the Seventieth Schedule
- 6.2 Where in Part I of the Seventieth Schedule a Parking Place is described as available for vehicles of a specified class the Driver of a vehicle shall not permit it to wait in that Parking Place unless it is of the specified class
- 6.3 The Driver of a vehicle shall not (except with the written consent of the Council) permit it to wait in any Parking Place in such a position that it causes or is likely to cause an obstruction to other users of the Parking Place or that it is not in accordance with the requirements of this Order
- 6.4 The Driver of a vehicle shall not (except with the written consent of the Council) permit it to wait in any Parking Place in a Parking Bay or Bays marked "Reserved" whether by a word or words painted on the ground or by any notice standing or otherwise displayed adjacent to the Parking Bay(s)
- 6.5 The Driver of a vehicle shall not permit it to wait in a Parking Bay that is marked in words or symbols as only available for the "disabled" unless there is displayed a Disabled Persons Badge only and the Disabled Person to whom the badge was issued accompanies the vehicle
- 6.6 The Driver of a vehicle shall not (except with the written consent of the Council) permit it to wait in a Parking Bay or Bays marked as being reserved for "High Vehicles" whether by a word or words painted on the ground or by any notice standing or otherwise displayed adjacent to the

Parking Bay(s) and for the purposes of this Order a High Vehicle shall be deemed to be a Vehicle which exceeds the height which enables it to pass clearly under any height barrier constructed by the Council in accordance with Article 7.2

- 6.7 A vehicle drawing a Trailer may wait in the Parking Place unless there is a sign in the Parking Place to the contrary. The provisions of this Order shall apply to a "Trailer" as they apply to a "vehicle" or "Authorised Vehicle" where the context allows
- 6.8 Subject to Article 6.9 an Authorised Vehicle must be parked wholly within a Parking Bay
- 6.9 In the event that due to the size of an Authorised Vehicle more than one Parking Bay is required the Driver must park the vehicle within the required number of Parking Bays only and pay the appropriate charges for each Parking Bay occupied in accordance with the provisions of this Order

7.0 COUNCIL'S POWERS

- 7.1 Nothing in Article 6.0 shall restrict the power of the Council to close to the public the whole or any part of the Parking Place either temporarily or permanently
- 7.2 Nothing in Article 6.0 shall restrict the power of the Council to construct height barriers in any Parking Place either temporarily or permanently

8.0 MAXIMUM PERIOD OF WAITING

8.1 The Driver of a vehicle shall not permit it to wait in the Parking Places known as Avon Run Road Southcliffe Road Steamer Point Highcliffe Cliff Top and Highcliffe Castle for longer than twenty four hours continuously between 01 October and 31 March inclusive without the prior written consent of the Council

- 8.2 Subject to Article 8.1 the Driver of a vehicle shall not permit it to wait in the Parking Places called Southcliffe Road Steamer Point and Highcliffe Castle overnight in accordance with times displayed at notice boards at each of these Parking Places
- 8.3 Between 01 April and 30 September inclusive the holder of a Season Ticket or Parking Permit shall not permit the vehicle to wait in the Priory Car Park for longer than four hours unless the holder has obtained the prior written consent of the Council
- 8.4 The Driver of a vehicle shall not permit the vehicle to wait in the Highcliffe Castle Car Park for longer than four hours without the prior written consent of the Council

9.0 CHARGES

9.1 General

Subject to Article 20.0 the Driver of an Authorised Vehicle shall pay the appropriate charge in accordance with the parking charges specified in Part I of the Seventieth Schedule provided that

- (a) the charge shall apply whether or not the Driver or passenger(s) remains with the vehicle within the Parking Place
- (b) where in a Parking Place it is indicated by notice that in respect of any day or any specified part of a day no charge will be made or such lesser charge than that specified in the Seventieth Schedule will be made then the charge (if any) shall be such lesser charge

10.0 SEASON TICKETS AND PARKING PERMITS

- 10.1 The Council may on an application of the Owner or Driver of an Authorised Vehicle issue a Season Ticket or Parking Permit in respect of that Authorised Vehicle on payment of the charge specified in Part II of the Seventieth Schedule
- 10.2 Subject to Article 7.0 a Season Ticket or Parking Permit so issued shall be valid in respect of such of the Parking Places described in Part II of the Seventieth Schedule as is specified thereof during the period for which it is issued

11.0 METHOD OF PAYMENT OF CHARGES

- 11.1 Subject to Article 11.3 the appropriate charge specified in Part I of Seventieth Schedule shall be paid by the Driver of an Authorised Vehicle leaving the Authorised Vehicle in a Parking Bay within a Parking Place by the insertion of an appropriate coin or coins into the Ticket Machine provided and where more coins than one are required they shall be inserted in the Ticket Machine immediately one after the other or where there are debit card or credit card facilities by inserting the appropriate debit card or credit card into the Ticket Machine provided
- 11.2 Each parking ticket to be purchased shall be such as may be necessary to show that the whole of the parking charge has been paid which charge is to be ascertained from the information shown on the Ticket Machine and/or notice board situated in the Parking Place
- 11.3 Each parking ticket purchased in accordance with this Article in payment of the parking charge as specified in Part I of the Seventieth Schedule shall be purchased immediately after the vehicle has stopped in the Parking Place for the purpose of waiting therein. Subject to any maximum period of waiting in force in respect of a Parking Place further tickets in payment

of the relevant parking charge for any subsequent period may be purchased at any time before the expiry of the period for which the parking charge has been paid

11.4 If at the time when a vehicle is left in the parking place there is no Ticket Machine at that Parking Place or all the Ticket Machines at that Parking Place carry notices upon them by a person duly authorised by the Council indicating that they are not in use or out of order the Driver of that vehicle shall pay the parking fee for use of that Parking Place on demand by any duly authorised officer or servant of the Council for which he shall receive a ticket or tickets. Article 12.0, 15.0 and 19.0 shall apply to any such ticket or tickets as though they were a ticket or tickets purchased in accordance with Article 9.0

12.0 DISPLAY OF TICKET

- 12.1 Any ticket or tickets issued including a Season Ticket or Parking Permit shall be displayed on the vehicle in respect of which it was issued in the following manner
 - (a) in the case of an Authorised Vehicle which is fitted with a transparent windscreen by sticking the ticket or tickets to the inside surface of the windscreen so that it is facing forwards and can be clearly seen from the front of the vehicle
 - (b) in the case of an Authorised Vehicle which is not fitted with a transparent windscreen by sticking the ticket or tickets on the nearside of the Authorised Vehicle not more than 2 metres and not less than 0.75 metres above the surface of the Parking Place in the immediate vicinity of the Authorised Vehicle so that it is facing forwards and the day and time of issue and other printed matter shown on the front of the ticket can be clearly seen from the outside of the vehicle

13.0 NON-TRANSFER OF TICKETS

- 13.1 It shall be an offence for the Driver of a vehicle using the Parking Place in Column (1) of Part I of the Seventieth Schedule to display any ticket other than the current ticket or tickets issued and each ticket purchased shall be valid on date of purchase only (except Mudeford Quay Mayors Mead and Avon Run Road where tickets are purchased for more than twenty four hours)
- 13.2 Each parking ticket shall be valid only for the vehicle for which it was purchased and shall not be valid if the ticket is transferred to another vehicle and in the event that the vehicle is parked in another Parking Place or Places on the same date as that shown on the parking ticket then the parking ticket shall not be valid in that Parking Place or Places

14.0 DAMAGED OR FOREIGN COINS

No person shall insert in any ticket machine in the Parking Place any bent or damaged or foreign coins by way of payment of the charge in respect of a vehicle

15.0 LIABILITY TO AND AMOUNT OF PENALTY CHARGE

- 15.1 Subject to Article 18.2 the Owner of a vehicle shall be liable to pay a Penalty Charge" in the following circumstances:-
 - (a) if an Authorised Vehicle is left in the Parking Place for longer than the period for which payment was made
 - (b) if an Authorised Vehicle is left in the Parking Place and no charge is paid
 - (c) if a vehicle other than an Authorised Vehicle is left in the Parking Place
 - (d) if an Authorised Vehicle is left in a Parking Place specified in Part I of the Seventieth Schedule and in respect of which a valid Season

Ticket or Parking Permit in accordance with Part II of the Seventieth Schedule has not been issued

- (e) if an Authorised Vehicle is left in any position other than that specified in Column (2) of Part I of the Seventieth Schedule or is causing an obstruction
- (f) if an Authorised Vehicle is left in a Parking Place and the ticket issued is not displayed in accordance with Article 12.0
- (g) if a vehicle which does not fall within the categories specified in Article 31.4(d) is parked in an area reserved for Disabled Drivers in accordance with Article 31.4(d)
- (h) if a vehicle is left in a Parking Place contrary to the requirements of Article 24.0
- (i) where there is a contravention of Articles 6.0, 8.0, 9.0, 13.0, 21.0, 22.0, 23.0, 25.0, 27.0, 29.0, 30.0 and 31.0
- 15.2 Where a vehicle has incurred a Penalty Charge a Parking Attendant shall attach to the vehicle in a conspicuous position a Penalty Charge Notice which shall include the following particulars
 - (a) the registration mark of the vehicle or where the vehicle is being used under a trade licence the number of trade plate carried by the vehicle
 - (b) the time at which the Parking Attendant observed that the Penalty Charge had been incurred
 - (c) the grounds on which the Parking Attendant believes that a Penalty Charge is payable with respect to the vehicle
 - (d) a statement that the Penalty Charge of Sixty Pounds (£60) is required to be paid
 - (e) that the Penalty Charge must be paid before the end of the period of 28 days beginning with the date of the Penalty Charge Notice
 - (f) that if the Penalty Charge is paid before the end of the period of 14 days beginning with the date of the Penalty Charge Notice, the amount of the Penalty Charge will be reduced to Thirty Pounds (£30)

- (g) that if the Penalty Charge is not paid before the end of the 28 day period, a Notice to Owner may be served by the Council on the person appearing to them to be the Owner of the vehicle
- (h) the address to which payment of the Penalty Charge must be sent

16.0 MAXIMUM PERIOD FOR WHICH AN AUTHORISED VEHICLE MAY BE LEFT AFTER PENALTY CHARGE INCURRED

- 16.1 Subject to Article 34.0 the period for which an Authorised Vehicle may be left in the Parking Place after the Penalty Charge has been incurred shall not exceed twelve hours and the Driver of the vehicle shall not leave it in the Parking Place for longer than that
- 16.2 If an Authorised Vehicle is left in the Parking Place for more than the said period after the Penalty Charge has been incurred it shall be the duty of a Parking Attendant to attach to the Authorised Vehicle in a conspicuous position a further Penalty Charge Notice which shall include the particulars mentioned in Article 15.2 and the following additional particulars namely the time at which he or she first noticed that the Authorised Vehicle had been left in the Parking Place for more than the prescribed period after the Penalty Charge had been incurred

17.0 RESTRICTIONS ON REMOVAL OF TICKETS AND PENALTY CHARGE NOTICES

- 17.1 When the ticket or tickets have been displayed on a vehicle in accordance with Article 12.0 above no person shall remove the ticket or tickets from the vehicle until the vehicle is removed from the Parking Place
- 17.2 Where a Penalty Charge Notice has been attached to a vehicle in accordance with the provisions of Articles 15.2 or 16.2 no person not being the Driver of the vehicle or person duly authorised by the Council shall

remove the Penalty Charge Notice from the vehicle unless authorised to do so by the Driver

18.0 MANNER OF PAYMENT OF PENALTY CHARGE

- 18.1 The Penalty Charge shall be paid to the Council:-
- (a) by post with payment being made by crossed cheque, bankers draft, money order or postal order made payable to 'Christchurch Borough Council' which shall be delivered to or sent by post to the Council's Resources Department at the Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ quoting the Penalty Charge Notice number;
- (b) in person by attending the Council Offices at the address above during office hours being 9.00am to 4.30pm Monday to Friday inclusive (excluding Bank Holidays) with payment being made by cash, crossed cheque, bankers draft, money order or postal order made to 'Christchurch Borough Council' or by credit or debit card;
- (c) by telephone call to the Council Offices on telephone number (01202)
 495252 with payment being made by credit or debit card;
- (d) on-line via the internet by visiting <u>www.dorsetforyou.com</u> and following the on-line instructions with payment being made by credit or debit card

Provided that if the Twenty-Eight day period expires upon a day on which the said Department is closed the period within which payment of the Penalty Charge shall be paid to the Council shall be extended until 1pm on the next full day on which the said Department is open

18.2 Provided that if within fourteen days following the date on which the Penalty Charge Notice is issued and the Driver tenders to the Council's Resources Department aforesaid the sum of Thirty Pounds (£30) then such sum or sums may be accepted by the Council in satisfaction of the Penalty Charge

19.0 INDICATIONS BY TICKET MACHINES AND TICKETS AS EVIDENCE

Unless the contrary is proved:-

- 19.1 If at any time an Authorised Vehicle is left in a Parking Place without any parking ticket displayed indicating payment of the appropriate charge in accordance with Article 11.0 it shall be presumed that no parking charge has been paid in respect of that vehicle
- 19.2 If at any time while an Authorised Vehicle is left in the Parking Place there is displayed on the Authorised Vehicle a ticket or tickets issued by a Ticket Machine indicating the day and the time of issue of the ticket or tickets and the time indicated on the Ticket Machine clock shows that the period for which the ticket or tickets are valid has expired it shall be presumed that the charge specified in Article 9.0 has been duly paid for the Authorised Vehicle and that the period for which the period for which the time period for which the payment was made by the charge has already expired
- 19.3 If at any time there is waiting in a Parking Place a vehicle whereon there is displayed one or more parking tickets but such ticket or tickets represent payment of a charge less than the charge prescribed in the Seventieth Schedule in respect of the Parking Place for the class to which the said vehicle belongs it shall be presumed that no parking charge has been paid in respect of the vehicle so waiting
- 19.4 If at any time there is waiting in a Parking Place a vehicle whereon there is displayed one or more parking tickets purchased from a Parking Place other than the Parking Place in which the vehicle is waiting it shall be presumed that no parking charge has been paid in respect of that vehicle
- 19.5 Without prejudice to Article 13.2 if at any time there is waiting in a Parking Place a vehicle whereon there is displayed one or more parking tickets or

a Parking Permit purchased for display on a vehicle other than the vehicle so waiting it shall be presumed that no parking charge has been paid in respect of that vehicle

- 19.6 If at any time while an Authorised Vehicle is left in the Parking Place there is displayed on the Authorised Vehicle a ticket or tickets issued by a Ticket Machine indicating the day and the time of expiry of the ticket or tickets and the time indicated on the Ticket Machine clock shows that the period for which the ticket or tickets are valid has expired more than twenty four hours later than the time on the ticket those indications shall be
 - (a) evidence that the vehicle was left in the Parking Place for longer after the Penalty Charge had been incurred than the time authorised by Article 16.1
 - (b) evidence that the charge was incurred for longer after the Penalty Charge had been incurred than the time authorised by Article 16.1

20.0 EXEMPTION FROM CHARGES

Except where expressly provided therein Articles 9.0 to 19.0 shall not apply to:-

- 20.1 pedal cycles and
- 20.2 solo motor cycles left in Parking Places provided they are parked so as not to occupy a Parking Bay or obstruct a Parking Bay or a pedestrian or vehicular access aisle
- 20.3 any vehicle being a vehicle for disabled persons and which is exempt from vehicle excise duty by virtue of paragraphs (18) or (19) of Schedule 2 to the Vehicles Excise and Registration Act 1994

21.0 SPEED OF VEHICLES

No person shall in the Parking Place drive or cause or permit to be driven any vehicle at a speed which exceeds ten miles per hour or such lesser speed as shall be reasonable according to the circumstances of the case at the time

22.0 RESTRICTION OF CLASS OF VEHICLE USING PARKING PLACE AND ON MANNER OF PARKING

- 2.2.1 Subject to Article 20.0 no vehicle other than a vehicle specified in Column(3) of Part I of the Fifty-Eighth Schedule shall be left in the Parking Place and no vehicle shall be left in a position other than that specified in Column(2) of Part I of the Seventieth Schedule
- 22.2 No person shall cause or permit a Mobile Home to be left in a Parking Place

23.0 DRIVERS TO SWITCH OFF ENGINE OF VEHICLE

The Driver of a vehicle using the Parking Place shall stop the engine as soon as the vehicle is in position in the Parking Bay and shall not start the engine except when about to change the position of the vehicle or to depart from a Parking Place

24.0 VEHICLES IN PARKING PLACE TO BE IN SATISFACTORY CONDITION

The Driver of a vehicle shall not permit the vehicle to wait in the Parking Place if the vehicle is in such condition as to be likely to cause a danger to other users of the Parking Place or damage to the surface condition of the Parking Place

25.0 DRIVER NOT TO SOUND THE HORN OF VEHICLE IN PARKING PLACE

The Driver of a vehicle using the Parking Place shall not sound any horn or other similar instrument in or attached to the vehicle except that the Driver may sound the horn in or attached to the vehicle for the purpose of giving any necessary warning when the vehicle is in motion

26.0 VEHICLE IN PARKING PLACE NOT TO BE DRIVEN EXCEPT FOR ENTERING OR LEAVING

No person shall except with the permission of any person duly authorised by the Council drive a vehicle in the Parking Place except in accordance with the provisions of this Order or for the purpose of entering and departing from the Parking Place and driving a vehicle around the Parking Place in accordance with the provisions of this Order for the purpose of locating an available Parking Bay

27.0 PROHIBITION OF LOUD SPEAKERS ETC

No person shall in the Parking Place operate any wireless loud speaker or similar device in or fixed or connected to a vehicle wantonly shout or otherwise make any loud noise likely to cause disturbance or annoyance to the users of the Parking Place or residents of premises in the neighbourhood

28.0 PROHIBITION OF BREACHES OF THE PEACE

No person shall in the Parking Place use any obscene threatening abusive or insulting language gesture or conduct with intent to put any person in fear or so as to occasion a breach of the peace or whereby a breach of the peace is likely to be occasioned

29.0 RESTRICTIONS ON THE USE OF A VEHICLE IN THE PARKING PLACE

No person shall use any part of the Parking Place or a vehicle left in the Parking Place

- 29.1 for camping purposes or overnight sleeping
- 29.2 for cooking or sanitary purposes

- 29.3 for the purpose of carrying out any work of washing maintenance or repair to a vehicle parked thereon except such as may be reasonably necessary to enable the vehicle to be moved from the Parking Place or
- 29.4 in connection with the sale of any article to any person in or near the Parking Place or in connection with the selling or offering for hire of any goods or of the skill or services of any persons unless authorised by the Council in writing

30.0 PROHIBITION OF PASSAGE THROUGH CAR PARK

No person shall use the Parking Place as a means of through passage for vehicles proceeding from one road to another or to a further length of the said road

31.0 DIRECTIONS FOR TRAFFIC TO BE FOLLOWED

When in the Parking Place signs are erected or surface markings are laid for the purpose of

- 31.1 Indicating the entrance to or the exit from the Parking Place or
- 31.2 Indicating that a vehicle using the Parking Place shall proceed in a specified direction within the Parking Place
- 31.3 Indicating an area reserved for Season Ticket or Parking Permit holder
- 31.4 Indicating an area reserved for Disabled Drivers

no person shall drive or cause or permit to be driven any vehicle

- (a) so that it enters the Parking Place otherwise than by an entrance or leaves the Parking Place otherwise than by an exit so indicated
- (b) in a direction other than that specified

- (c) so that it enters the area reserved for Season Ticket or Parking Permit holders unless it is an authorised vehicle in respect of which a Season Ticket or Parking Permit has been issued and a Season Ticket or Parking Permit is displayed in accordance with Article 12.0 as the case may be or
- (d) so that it enters the area reserved for Disabled Drivers unless the vehicle is either
 - (i) an Invalid Carriage or
 - (ii) an Authorised Vehicle left by the Driver thereof who is a person to whom there has been issued a Disabled Persons Badge provided such a badge is displayed on the vehicle in a position as required by the Regulations as aforesaid

32.0 RESTRICTIONS ON USE OF THE PARKING PLACE

Without prejudice to the other provisions of this Order when in a Parking Place no person shall

- 32.1 Erect or cause or permit to be erected any tent booth stand building skip stall or other structure for any purpose whatsoever or deposit materials or excavate any surfaces without the prior written consent of the Council
- 32.2 Light or cause or permit to be lit any fire or barbecue
- 32.3 Carry out any trade or business activity or charitable activity without the prior written consent of the Council
- 32.4 Affix or attach posters bills or any advertising material whatsoever to either the walls or structure of the Parking Place or to any vehicle parked in the Parking Place or distribute leaflet or advertising material of any kind
- 32.5 Ride or use a skateboard roller skates or boots or similar thing or things

- 32.6 Play or participate in any formal or organised game sport or recreation in a Parking Place without the prior written consent of the Council
- 32.7 Use the Parking Place for the purpose of campaigning for political or other purposes
- 32.8 Damage or deface any furniture sign surface or structure or vehicle by means of graffiti or other action
- 32.9 Hold or take part in any organised picnic without the prior written consent of the Council
- 32.10 Carry out or permit to be carried out to the vehicle any work of cleaning construction alteration or maintenance or repair except such repair as may be necessary to enable the vehicle to be moved from the Parking Place
- 32.11 Cause or permit a vehicle to be parked in such a manner in a Parking Place as to cause or be likely to cause an obstruction to other users of the Parking Place
- 32.12 Wantonly shout make any loud noise or otherwise cause a disturbance or annoyance to users of the Parking Place or residents of premises in the neighbourhood
- 32.13 Cause or permit to be thrown down dropped or otherwise deposited and left any waste litter or detritus

33.0 APPLICATION OF ENACTMENT

Sub-Sections (4) (5) and (6) of Section 47 of the Act shall apply to the Parking Places as if it was a Parking Place provided on the highway

34.0 REMOVAL ETC OF VEHICLES FROM A PARKING PLACE

- 34.1 When a Parking Attendant is of the opinion that any of the provisions of Articles 6.0, 8.0, 9.0, 12.0, 13.0, 16.0, 22.0, 24.0, 32.10 and 32.11 have been contravened in respect of any vehicle left in the Parking Place he may remove the said vehicle or arrange for it to be removed from the said Parking Place
- 34.2 When an Authorised Vehicle is waiting in the Parking Place in contravention of Article 22.0 any person duly authorised by the Council may alter or cause to be altered the position of the vehicle in order that its position shall comply therewith
- 34.3 A Parking Attendant may move or cause to be moved in case of emergency to any place he or she thinks fit any vehicle left in the Parking Place

35.0 MANNER OF REMOVAL OF VEHICLES

Any person removing or altering the position of a vehicle by virtue of Article 34.0 may do so by towing or driving the vehicle or in such other manner as he or she may think necessary and may take such measures in relation to the vehicle as he or she may think necessary to enable him or her to remove it or alter its position as the case may be

36.0 SAFE CUSTODY OF VEHICLES REMOVED FROM THE PARKING PLACE

When a person authorised by the Council removes or makes arrangements for the removal of a vehicle from the Parking Place by virtue of Article 34.0 he or she shall make such arrangements as may be reasonably necessary for the safe custody of the vehicle following such removal Section 101 and 102 of the Act shall be applicable by the Council in respect of the vehicle insofar as the Council shall desire

 THE COMMON SEAL of THE
)

 BOROUGH COUNCIL OF
)

 CHRISTCHURCH was hereunto
)

 affixed in the presence of:)

Borough Solicitor

THE FIRST SCHEDULE NO WAITING AT ANY TIME

Abingdon Drive Addiscombe Road Airfield Road Airfield Way Amble side Amethyst Road Apple Close Argyle Road Arthur Lane Arthur Road Asquith Close Avenue Road Avon Buildings Avon Run Road Avon View Road **Bailey Drive Bargates** Bargates Co-op Access Road **Bargates Service Road** Barrack Road **Beaconsfield Road Beaulieu Avenue** Beaulieu Road **Belvedere Road Bosley Close** Bosley Way Brabazon Drive Braemar Drive **Bridge Street Bronte Avenue Bub Lane Buccaneers Close Buce Haves Close** Bure Lane **Burley Road Burnett Avenue Burnett Road** Burnham Road **Burton Road** The Buttery Cameron Road Campbell Road **Castle Avenue** Castle Street **Chapel Gate** Chapel Lane Chestnut Way Chewton Common Road

Chichester Way Church Lane Church Street Clarendon Road Coastguard Way Court Close Cricket Close Croft Road Cross Way Dairy Close De Havilland Way Dennistoun Avenue Devon Road Disraeli Road Dorset Road **Douglas Avenue** Dragoon Way **Ducking Stool Lane** Earlsdon Wav Edward Road Elm Avenue Elizabeth Avenue Essex Avenue Fairfield Fairmile Road Farm Lane Fitzmaurice Road Fountain Way Gardner Road **Gleadowe Avenue** Glenville Road Gordon Road Grange Road Greenways The Grove Groveley Road Grove Road West Harbour Crescent Heather Close High Street Highview Close Hillside Drive Hinton Wood Avenue Holme Road Hunt Road Hurn Court Lane Hurn Road

Jumpers Avenue Jumpers Road **Kings Avenue** Latch Farm Avenue Ledbury Road Lights Close Livingstone Road Lodge Road Lymington Road Magdalen Lane Manor Road Marlow Drive Marlpit Drive The Meridians Merritown Lane Mill Lane Mill Road Millhams Street Millhams Street North Moffat Road Monkswell Green Montagu Road Mudeford **Mudeford Green Close** Mudeford Lane Mude Gardens Nea Road Newlands Road Normandy Drive Oak Avenue **Orchard Close Orchid Way** Park Close Parley Green Lane Parley Lane Pauntlev Road Pennant Road Pinewood Close Poplar Close Portfield Road **Princess Avenue** Purewell Purewell Close Purewell Cross Road Quay Road **Queens Avenue** Queens Road **Regency Close Regency Crescent** Regent Way Reid Street Ringwood Road **Riverlea Road**

River Way Rothesay Drive Rotterdam Drive **Royal Close Russell Drive** Salisbury Road Sandown Road Sandy Lane St. Margarets Avenue Saufland Drive Scotts Hill Lane Seaton Road Silver Street Smugglers Lane North Somerford Road Somerford Rd Service Road Somerford Way Somerset Road Sopers Lane Southcliffe Road South View Road Stanley Road Stanpit Station Road Stony Lane Stony Lane South Stour Road Stourbank Road Stourvale Avenue Stour Way Stroud Park Avenue Stuart Road Suffolk Avenue Tresillian Way Twynham Avenue The Lawns Valiant Wav Victoria Road Viking Way Warren Avenue Waterford Road Watermill Road Wharncliffe Gardens Wharncliffe Road Whitehall Wickfield Avenue Wickfield Close Wick Lane Willow Drive Willow Way Wilverley Road Woodland Way Wortley Road

THE SECOND SCHEDULE

NO WAITING AT ANY TIME 01 APRIL TO 30 SEPTEMBER

Mudeford Lane

THE THIRD SCHEDULE

NO WAITING AT ANY TIME 01 MAY TO 30 SEPTEMBER

Fairway Drive Stour Way Whitehall Wick Lane

THE FOURTH SCHEDULE

NO WAITING AT ANY TIME 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM

Abbotts Close Anchor Close Arundel Way Avon Run Close Avon Run Road **Beacon Drive** Brook Way **Bure Haven Drive Bure Homage Gardens Bure Close Bure Lane Bure Road** Capesthorne Castle Avenue Copse Way **Dunlin Close** East Cliff Way Elmwood Way Farm Lane Falcon Drive Freshwater Road Friars Road Fulmar Road **Glengarry Way** Hinton Wood Avenue Hynesbury Road Island View Avenue Lymington Rd Service Road Maple Close Medina Way

Merlin Way Montagu Road Mudeford Nea Road **Oakleigh Way** Peregrine Road Pine Crescent **Priory Close** Ranelagh Road Raven Way Redan Close Robins Way Rothesay Drive Saxonford Road Seafield Road Seaway Avenue Sheldrake Road Silver Way Smugglers Lane North **Smugglers Lane South** Southcliffe Road Stuart Road The Lawns Vecta Close Waterford Gardens Waterford Road Wharncliffe Gardens Wharncliffe Road Woodland Way Wren Close

THE FIFTH SCHEDULE

NO WAITING AT ANY TIME 01 APRIL T0 30 SEPTEMBER 10.00 PM TO 7.00 AM

Stour Way

THE SIXTH SCHEDULE

NO WAITING 10.00 AM TO 6.00 PM

Bargates Bridge Street Buce Hayes Close Chewton Common Road Elphinstone Road Gordon Road Heather Close Purewell Purewell Close St Margarets Avenue Stanley Road Stony Lane Wickfield Avenue Wickfield Close Wortley Road

THE SEVENTH SCHEDULE

NO WAITING 10.00 AM TO 7.00 PM

Falcon Drive Service Road Waterford Place

THE EIGHTH SCHEDULE

NO WAITING 8.00 AM TO 6.00 PM

Airfield Way High Street Buce Hayes Close
THE NINTH SCHEDULE

NO WAITING 9.00 AM TO 5.00 PM

Buccaneers Close Wick Lane

THE TENTH SCHEDULE

NO WAITING 9.00 AM TO 6.00 PM

Barrack Road Fairfield Groveley Road

THE ELEVENTH SCHEDULE

NO WAITING 9.00 AM TO 7.00 PM

Chewton Common Road

THE TWELTH SCHEDULE

NO WAITING MONDAY TO FRIDAY 10.00 AM TO 4.00 PM

Quay Road

THE THIRTEENTH SCHEDULE

NO WAITING MONDAY TO FRIDAY 10.00 AM TO 6.00 PM

Caxton Close Edward Road Newlands Road Somerford Road Service Road Southey Road

THE FOURTEENTH SCHEDULE

NO WAITING MONDAY TO FRIDAY 8.30 AM TO 4.30PM

Pauntley Road Queens Road

THE FIFTEENTH SCHEDULE

NO WAITING MONDAY TO FRIDAY 8.00 AM TO 6.00 PM

Airfield Road

THE SIXTEENTH SCHEDULE

NO WAITING MONDAY TO FRIDAY 9.00 AM TO 4.00 PM

Mill Road

THE SEVENTEENTH SCHEDULE

NO WAITING MONDAY TO SATURDAY 8.30 AM TO 9.30 AM AND 3.00 PM TO 4.00 PM

Dorset Road

THE EIGHTEENTH SCHEDULE

NO WAITING MONDAY TO FRIDAY 10.00 AM TO 12.00 (NOON)

Greywell

THE NINETEENTH SCHEDULE

NO WAITING MONDAY TO SATURDAY 10.00 AM TO 6.00 PM

Heather Close Poplar Close Seaton Road Twynham Avenue

THE TWENTIETH SCHEDULE

WAITING LIMITED DISABLED ONLY 10.00 AM TO 4.00 PM THREE HOURS NO RETURN WITHIN ONE HOUR

High Street

THE TWENTY-FIRST SCHEDULE

WAITING LIMITED TO 10.00 AM TO 6.00 PM TWO HOURS NO RETURN WITHIN TWO HOURS

Lymington Road

THE TWENTY-SECOND SCHEDULE

WAITING LIMITED TO 10.00 AM TO 6.00PM TWO HOURS NO RETURN WITHIN 30 MINUTES

Mudeford Lane

THE TWENTY-THIRD SCHEDULE

WAITING LIMITED TO 10.00 AM TO 6.00 PM TO 30 MINUTES NO RETURN WITHIN 30 MINUTES

Bridge Street Mudeford Riverlea Road Somerford Road Wortley Road

THE TWENTY-FOURTH SCHEDULE

WAITING LIMITED TO 10.00 AM TO 6.00 PM FOR 30 MINUTES NO RETURN WITHIN ONE HOUR

High Street Stour Road Wick Lane

THE TWENTY-FIFTH SCHEDULE

WAITING LIMITED TO 10.00 AM TO 6.00 PM FOR 30 MINUTES IN ANY HOUR

Bargates Bridge Street Ringwood Road

THE TWENTY-SIXTH SCHEDULE

WAITING LIMITED TO 10.00 AM TO 6.00 PM FOR ONE HOUR

Gordon Road

THE TWENTY-SEVENTH SCHEDULE

WAITING LIMITED TO 10.00 AM TO 7.00 PM FOR ONE HOUR NO RETURN WITHIN ONE HOUR

Waterford Road

THE TWENTY-EIGHTH SCHEDULE

WAITING LIMITED TO 10.00 AM TO 7.00 PM FOR 30 MINUTES NO REUTRN WITHIN 30 MINUTES

Falcon Drive Service Road

THE TWENTY-NINTH SCHEDULE

WAITING LIMITED TO TWO HOURS NO RETURN WITHIN ONE HOUR

Wharncliffe Road

THE THIRTIETH SCHEDULE

WAITING LIMITED TO TWENTY-FOUR HOUR LIMIT

Cameron Road Wickfield Avenue

THE THIRTY-FIRST SCHEDULE

WAITING LIMITED TO TWENTY-FOUR HOURS NO RETURN WITHIN ONE HOUR NO SLEEPING BETWEEN HOURS OF 9.00 PM TO 6.00 AM

Ledbury Road

THE THIRTY-SECOND SCHEDULE

WAITING LIMITED TO 30 MINUTES IN ANY ONE HOUR

Coastguard Way

THE THIRTY-THIRD SCHEDULE

WAITING LIMITED TO 30 MINUTES

Christchurch Bypass (South Side)

THE THIRTY-FOURTH SCHEDULE

WAITING LIMITED TO 60 MINUTES NO RETURN WITHIN 60 MINUTES

Barrack Road

THE THIRTY-FIFTH SCHEDULE

WAITING LIMITED TO 8.00 AM TO 6.00 PM FOR 30 MINUTES NO RETURN WITHIN 30 MINUTES

Arthur Road

THE THIRTY-SIXTH SCHEDULE

PERMIT PARKING ONLY

Ducking Stool Lane Millhams Street

THE THIRTY-SEVENTH SCHEDULE

DISABLED PARKING ONLY

Stanpit Livingstone Road

THE THIRTY-EIGHTH SCHEDULE

WAITING LMITED TO 9.00 AM TO 6.00 PM 30 MINUTES NO RETURN WITH 30 MINUTES

Lymington Road

THE THIRTY-NINTH SCHEDULE

WAITING LIMITED 01 APRIL TO 30 SEPTEMBER 24 HOURS NO RETURN WITHIN ONE HOUR

Peregrine Road

THE FORTIETH SCHEDULE

WAITING LIMITED 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM ONE HOUR

Capesthorne

THE FORTY-FIRST SCHEDULE

WAITING LIMITED 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM 60 MINUTES NO RETURN WITHIN 30 MINUTES

Falcon Drive

THE FORTY-SECOND SCHEDULE

WAITING LIMITED MONDAY TO FRIDAY 10.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR

Caxton Close

THE FORTY-THIRD SCHEDULE

WAITING LIMITED MONDAY TO FRIDAY 10.00 AM TO 4.00 PM TWO HOURS NO RETURN WITHIN ONE HOUR

Stanley Road

THE FORTY-FOURTH SCHEDULE

WAITING LIMITED MONDAY TO FRIDAY 9.00 AM TO 5.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR (EXCEPT PERMIT HOLDERS)

Grange Road
THE FORTY-FIFTH SCHEDULE

WAITING LIMITED MONDAY TO FRIDAY 8.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR (EXCEPT PERMIT HOLDERS)

Riverlea Road

THE FORTY-SIXTH SCHEDULE

WAITING LIMITED MAY TO SEPTEMBER 10.00 AM TO 6.00 PM 30 MINUTES NO RETURN WITHIN ONE HOUR

Queens Avenue Whitehall

THE FORTY-SEVENTH SCHEDULE

WAITING LIMITED MONDAY TO SATURDAY 10.00 AM TO 6.00PM ONE HOUR NO RETURN WITHIN ONE HOUR

Stanpit

THE FORTY-EIGHTH SCHEDULE

WAITING LIMITED MONDAY TO SATURDAY 8.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR

Stour Road

THE FORTY-NINTH SCHEDULE

WAITING LIMITED MONDAY TO SATURDAY 10.00 AM TO 6.00 PM THIRTY MINUTES NO RETURN WITHIN THIRTY MINUTES

Purewell

THE FIFTIETH SCHEDULE

WAITING LMITED MONDAY TO SATURDAY 8.00 AM TO 6.00 PM ONE HOUR NO RETURN WITHIN ONE HOUR (EXCEPT PERMIT HOLDERS)

Avon Buildings Fairfield Princess Avenue Quay Road Station Road Twynham Avenue

THE FIFTY-FIRST SCHEDULE

WAITING LIMITED MONDAY TO SATURDAY 8.00 AM TO 6.00 PM TWO HOURS NO RETURN WITHIN TWO HOURS

St Catherines Parade

THE FIFTY-SECOND SCHEDULE

WAITING LIMITED MONDAY TO SATURDAY 9.00 AM TO 5.00 PM SIXTY MINUTES NO RETURN WITHIN SIXTY MINUTES (EXCEPT PERMIT HOLDERS)

Beaconsfield Road

THE FIFTY-THIRD SCHEDULE

WAITING LIMITED MONDAY TO SATURDAY 9.00 AM TO 6.00 PM SIXTY MINUTES NO RETURN WITHIN SIXTY MINUTES

Barrack Road

THE FIFTY-FOURTH SCHEDULE

CLEARWAY (NO STOPPING)

Christchurch By-Pass Lyndhurst Road Roeshot Hill

THE FIFTY-FIFTH SCHEDULE

COACHES ONLY WAITING LIMITED FIFTEEN MINUTES NO RETURN WITHIN THIRTY MINUTES

High Street Wick Lane

THE FIFTY-SIXTH SCHEDULE

NO LOADING AT ANY TIME

Bargates Bargates Service Road Castle Street Ducking Stool Lane High Street Millhams Street North Quay Road Sandown Road Somerford Road Sopers Lane

THE FIFTY-SEVENTH SCHEDULE

NO LOADING 10.00 AM TO 3.00 PM

Buce Hayes Close Lymington Road

THE FIFTY-EIGHTH SCHEDULE

NO LOADING 8.00 AM TO 5.00 PM

Wick Lane Wickfield Avenue

THE FIFTY-NINTH SCHEDULE

NO LOADING AT ANY TIME EXCEPT TO NUMBER 2 SOPERS LANE

Sopers Lane

THE SIXTIETH SCHEDULE

NO LOADING AT ANY TIME EXCEPT TO PREMISES IN CHURCH STREET

Church Lane Church Street

THE SIXTY-FIRST SCHEDULE

NO LOADING AT ANY TIME EXCEPT TO NUMBERS 19, 21, 23, 25, 30 AND 32 HIGH STREET

High Street

THE SIXTY-SECOND SCHEDULE

NO LOADING 8.00 AM TO 6.00 PM EXCEPT BY COMMERCIAL VEHICLES TO 45 – 67 ODDS, 46 – 76 EVENS HIGH STREET

High Street

THE SIXTY-THIRD SCHEDULE

NO LOADING AT ANY TIME EXCEPT TO CHURCH FOR WEDDINGS AND FUNERALS

Millhams Street

THE SIXTY-FOURTH SCHEDULE

NO LOADING AT ANY TIME EXCEPT TO 26 AND 27 MILLHAMS STREET

Millhams Street

THE SIXTY-FIFTH SCHEDULE

NO LOADING 01 APRIL TO 30 SEPTEMBER 10.00 AM TO 7.00 PM

Avon Run Road Southcliffe Road

THE SIXTY-SIXTH SCHEDULE

GOODS VEHICLES LOADING ONLY MONDAY TO SATURDAY 8.00 AM TO 6.00PM 20 MINUTE LIMIT

Wick Lane

THE SIXTY-SEVENTH SCHEDULE

BUS STOP CLEARWAY 7.00 AM TO 7.00 PM (NO STOPPING)

Bargates **Barrack Road** Bridge Street **Bronte Avenue Bure Haven Drive** Bure Lane **Burley Road** Burnham Road **Burton Road** Campbell Road Castle Avenue Castle Street De Havilland Way Dorset Road Edward Road **Everest Road** Fairmile Road Forest Rise Forest Way Fountain Way **Gleadowe Avenue** The Grove **Highcliffe Road** High Street Hinton Wood Avenue Hoburne Lane Hunt Road Hurn Road **Kings Avenue** Langley Road Lymington Road Lyndhurst Road Marsh Lane Martins Hill Lane

The Meadway Moorcroft Avenue Mudeford Mudeford Lane Normandy Drive Oak Avenue Peregrine Road Pittmore Road Purewell **Purewell Cross Road Ringwood Road River Way** The Runway Sainsburys Access Road St. Catherines Way St. Margarets Avenue Salisbury Road Sandown Road Smugglers Lane North Somerford Road Sopers Lane Stanpit Stony Lane Stour Road Stourvale Avenue Suffolk Avenue Walkford Road Wick Lane Willow Way Woodfield Gardens Woodhayes Avenue Woodstock Road

THE SIXTY-EIGHTH SCHEDULE

HACKNEY CARRIAGE STAND (TAXIS) (NO STOPPING)

Bargates Bargates Service Road High Street Lymington Road Somerford Road Wick Lane

THE SIXTY-NINTH SCHEDULE

NO STOPPING MONDAY TO FRIDAY 8.00 AM TO 5.00 PM ON SCHOOL ENTRANCE MARKINGS

Addiscombe Road Bingham Road Chewton Common Road Clarendon Road Draper Road Greenways Marmion Green Mudeford Lane Parkside Pauntley Avenue Portfield Road Queens Road Salisbury Road Sopers Lane Woodfield Gardens

THE SEVENTIETH SCHEDULE PART I PARKING CHARGES FOR OFF-STREET PARKING From 01 April 2007 until further notice

(All charges inclusive of VAT where applicable)

(1)	(2)	(3)			CHARGES -	(4) – SUBJECT TO	O ARTICLES 8	AND 9		
NAMÉ OF PARKING PLACE	POSITION IN WHICH VEHICLES MAY WAIT	CLASS OF VEHICLE PERMITTED	Class of Vehicle	Charge for up to 30 mins parking	Charge for up to 1 hour parking	Charge for up to 2 hours parking	Charge for up to 3 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking	Charge for any period between 7pm & 8am
Saxon Square	Wholly within marked	Motor Cars, Invalid Carriages	(a) All vehicles except Goods Vehicles	30p	60p	£1.00	£1.80	£2.50	£5.20	No Charge
Car Park	parking bay	and Goods Vehicles	(b) Goods Vehicles	90p	£1.80	£3.00	£5.40	£7.50	£15.60	
Bank Close Car Park	nk Close Wholly within Invalid Carriages	Motor Cars, Invalid Carriages and Goods	(a) All vehicles except Goods Vehicles	N/A	60p	£1.00	£1.80	£2.50	£5.20	No Charge
CarPark	parking bay	Vehicles	(b) Goods Vehicles	N/A	£1.80	£3.00	£5.40	£7.50	£15.60	
Bridge Street Car Park	Wholly within marked	Motor Cars, Invalid Carriages and Goods	(a) All vehicles except Goods Vehicles	N/A	60p	£1.00	£1.80	£2.50	£5.20	No Charge
Car Park	parking bay	Vehicles	(b) Goods Vehicles	N/A	£1.80	£3.00	£5.40	£7.50	£15.60	
By Pass	Wholly within	Motor Cars, Invalid Carriages,	(a) All vehicles except Goods Vehicles	N/A	60p	£1.00	£1.80	£2.50	£5.00	No Charge
Car Park Inalkeu	parking bay	Trailers including Caravans and Goods Vehicles	(b) Goods Vehicles	N/A	£1.80	£3.00	£5.40	£7.50	£15.00	
Wick Lane	Wholly within marked	Motor Cars, Invalid Carriages and Goods	(a) All vehicles except Goods Vehicles	N/A	60p	£1.00	£1.80	£2.50	£5.20	No Charge
Car Park	parking bay	Vehicles	(b) Goods Vehicles	N/A	£1.80	£3.00	£5.40	£7.50	£15.60	

(1)	(2) POSITION IN WHICH	(3) CLASS OF VEHICLE			CHARGES - SI	(4) JBJECT TO ART	ICLES 8 AND 9			
NAME OF PARKING PLACE	VEHICLES MAY WAIT	PERMITTED	Class of vehicle	Charge for up to 1 hour	Charge for up to 2 hours parking	Charge for up to 3 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking	Charge for any period between 7pm and 8am	
Pit Site Car Park	Wholly within marked parking bay or in a position which will not impede access to or exit from car park or otherwise cause an obstruction	Motor Cars, Invalid Carriages, Trailers including Caravans, Goods Vehicles, Heavy Commercial Vehicles and	 (a) All vehicles except Goods Vehicles and Heavy Commercial Vehicles which must pay three times the standard rate (b) Public 	60p	£1.00	£1.80	£2.50	£5.00	No Charge	
		Coaches	Service Vehicles	No Charge						
Wortley Road Car Park	Wholly within marked parking bay	Motor Cars, Invalid Carriages and	(a) All vehicles except Goods Vehicles	50p	£1.00	£1.50	£2.00	£4.00	No Charge	
		Goods Vehicles	(b) Goods Vehicles	£1.50	£3.00	£4.50	£6.00	£12.00		
Bargates (Law Courts)	Wholly within marked parking bay	Motor Cars, Invalid Carriages and	(a) All vehicles except Goods Vehicles	60p	£1.00	£1.50	£2.00	£4.00	No Charge	
Čar Park		Goods Vehicles	(b) Goods Vehicles	£1.80	£3.00	£4.50	£6.00	£12.00		

(1) NAME OF PARKING PLACE	(2) POSITION IN WHICH VEHICLES MAY WAIT	(3) CLASS OF VEHICLE PERMITTED	(4) CHARGES – SUBJECT TO ARTICLES 8 AND 9					
			Class of vehicle	Charge for up to 2 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking	Charge for any period between 7pm and 8am	
Willow Way Car Park	Wholly within marked parking bay		(a) All vehicles except Goods Vehicles	60p	£1.20	£3.00		
		excluding Caravans and Goods Vehicles	(b) Goods Vehicles	£1.80	£3.60	£9.00	No Charge	
Leisure Centre Car Park	Wholly within marked parking bay or where appropriate in a position which will not impede access to or exit from the car park or Leisure Centre or	Carriages, Trailers excluding Caravans, Goods Vehicles, Heavy Commercial	(a) All vehicles except Goods Vehicles which must pay three times the standard charge	60p	£1.20	£3.00	No Charge	
	otherwise cause an obstruction	Vehicles and Coaches	(b) Public Service Vehicles		No Charge			
Purewell Dairy Car Park	Wholly within marked parking bay		(a) All vehicles except Goods Vehicles	60p	£1.20	£3.00	No Charge	
		Vehicles	(b) Goods Vehicles	£1.80	£3.60	£9.00	(between 5pm and 8am)	

(1) NAME OF PARKING PLACE	(2) POSITION IN WHICH VEHICLES MAY WAIT	(3) CLASS OF VEHICLE PERMITTED	(4) CHARGES – SUBJECT TO ARTICLES 8 AND 9									
			Class of vehicle	Charge for up to 1 hour parking	Charge for up to 2 hours parking	Charge for up to 3 hours parking	Charge for up to 4 hours parking	Charge for up to 8 hours parking	Charge for up to 24 hours parking			
					Peri	od from 01	April to 31	Мау				
Mudeford Quay Car Park	In marked spaces but otherwise in a position which will not	Motor Cars, Invalid Carriages, Trailers	All vehicles except Goods	£1.00	£1.80	£2.60	£3.20	£6.00	£9.00			
Carraik	impede access to or exit from the	including Caravans and Goods Vehicles	including Caravans	Vehicles which		Fro	om 01 June	e to 31 Octo	ber	<u>I</u>		
	car park or otherwise cause an obstruction except that vehicles			and Goods Vehicles	must pay three times the	£1.20	£2.20	£3.20	£4.20	£8.00	£12.00	
	shall not wait in that part of the		standard charge		Fr	om 01 Nov	ember to 2	9 February	1			
	Quay that extends south- westwards from a point marked					60p	£1.00	£1.60	£2.20	£4.00	£6.00	
	by the easternmost end of the			From 01 March to 31 March					<u>I</u>			
	Highcliffe Sailing Club and easternmost end of the Haven			£1.00	£1.80	£2.60	£3.20	£6.00	£9.00			
	House Inn. Parking is not permitted within the Boat Park.				(up to 16		/ Ticket m time of p	L				
										er	£30.0 £50.0	00 per weel 00 per weel 00 per weel
					i – 31 Marc			00 per wee 00 per wee				

(1) NAME OF PARKING PLACE	(2) POSITION IN WHICH VEHICLES MAY WAIT	(3) CLASS OF VEHICLE PERMITTED	(4) CHARGES – SUBJECT TO ARTICLES 8 AND 9					
			Class of Vehicle	Charge for up to 2 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking		
Christchurch Quay				Period	from 01 April to 3	1 October		
Car Park				£1.70	£2.50	£5.00		
				From 0 ²	November to 29	February		
				£1.20	£2.00	£3.00		
				From	n 01 March to 31	March		
				£1.70	£2.50	£5.00		
Priory	Wholly within marked		All vehicles	Period from 01 April to 30 September				
Car Park	parking bay	Carriages		£1.70	£2.50	£5.00		
				From	01 October to 31	March		
				£1.20	£2.00	£3.00		
				From	n 01 March to 31	March		
				£1.70	£2.50	£5.00		

(1) NAME OF PARKING PLACE	INÉ OF POSITION IN WHICH CLASS OF VEHICLE CHARGE RKING VEHICLES MAY WAIT PERMITTED LACE				(4) S – SUBJECT TO ARTICLES 8 AND 9					
			Class of vehicle	Charge for up to 2 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking	Charge for up to 24 hours parking	Charge for up to 36 hours parking		
Mayors Mead Wholly within marked parking		Motor Cars, Invalid	(a) All vehicles except Goods Vehicles which must	Period from 01 April to 31 October						
Car Park	Car Park bay Ca	Carriages, Trailers including Caravans and Coaches	luding Caravans charge	£1.70	£2.50	£5.00	£10.00	£15.00		
			(b) Public Service Vehicles	No Charge						
	Trailer Park included in,	(a) All vehicles except GTrailers excludingVehicles which must pay		s From 01 November to 29 February						
	adjacent to and west of main car park and additionally	Caravans	three times the standard charge	£1.20	£2.00	£3.00	£6.00	£9.00		
	where parked in area on east side of car park designated by		(b) Public Service Vehicles	No Charge						
notices			a) All vehicles except Goods Vehicles which must pay three times the standard charge	From 01 March to 31 March						
			(b) Public Service Vehicles	£1.70	£2.50	£5.00	£10.00	£15.00		

(1) NAME OF PARKING PLACE	(2) POSITION IN WHICH VEHICLES MAY WAIT	(3) CLASS OF VEHICLE PERMITTED					
			Class of vehicle	Charge for up to 1 hour parking	Charge for up to 2 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking
Avon Run Road	Wholly within marked	Motor Cars, Invalid	All vehicles except			April to 31 May	
Car Park	parking bay	Carriages, Trailers excluding		80p	£1.30	£2.40	£4.00
		Caravans and Goods	which must pay	From 01 June to 31 October			
		vehicles	three times the standard charge	£1.00	£1.80	£3.50	£5.00
				From 01 November to 29 February			
						harge	
						h to 31 March	
				80p	£1.30	£2.40	£4.00
				(up to		/ Ticket m time of purch	nase)
				01 April – 31 M			per week
				01 June - 31 0			per week
				01 March – 31	March	£15.00	, per week
	Wholly within marked		a) All vehicles		From 01 Ap	ril to 31 May	
	parking bay or in a position	Carriages, Trailers including		80p	£1.30	£2.40	£4.00
overspill area	which will not impede access		Vehicles and Heavy			to 31 October	
	to or exit from the car park or			90p	£1.70	£3.50	£5.00
	otherwise cause an	Coaches	Vehicles which	Fro		er to 29 Februa	ary
	obstruction		must pay three times the Standard			harge	
			charge			h to 31 March	
				80p	£1.30	£2.40	£4.00
			b) Public Service		From 01 Apri		
			Vehicles		No C	harge	

(1) NAME OF PARKING PLACE	NAME OF POSITION IN WHICH CLASS OF VEHICLE RKING PLACE VEHICLES MAY WAIT PERMITTED				(4) CHARGES – SUBJECT TO ARTICLES 8 AND 9					
			Class of vehicle	Charge for up to 1 hour parking	Charge for up to 2 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking			
		t Motor Cars, Invalid Carriages	All vehicles	F	Period from 01	April to 31 Mag	y			
Car Park	otherwise in a position which will not impede access to or			80p	£1.30	£2.40	£4.00			
c	exit from the car park or				From 01 June	to 31 October	1			
	otherwise cause an obstruction			£1.00	£1.80	£3.50	£5.00			
				From 01 November to 29 February						
				No Charge						
					01 March t	o 31 March				
				80p	£1.30	£2.40	£4.00			
Steamer Point	Wholly within marked	Motor Cars and Invalid	All vehicles	From 01 April to 31 May						
Car Park	parking bay or in a position which will not impede access			80p	£1.30	£2.40	£4.00			
	to or exit from the car park or			From 01 June to 31 October						
	otherwise cause an obstruction			£1.00	£1.80	£3.50	£5.00			
				From 01 November to 29 February						
				No Charge						
				From 01 March to 31 March						
				80p	£1.30	£2.40	£4.00			

(1) NAME OF PARKING PLACE	(2) POSITION IN WHICH VEHICLES MAY WAIT	(3) CLASS OF VEHICLE PERMITTED	(4) CHARGES – SUBJECT TO ARTICLES 8 AND 9		8 AND 9			
Highcliffe Castle Car Park	Wholly within marked parking bay or in a position which will not impede access to or exit	Invalid Carriages	Class of vehicle	Charge for up to 1 hour parking	Charge for up to 2 hours parking	Charge for up to 3 hours parking	Charge for up to 4 hours parking	Charge for up to 12 hours parking
	form the car park or otherwise cause an obstruction		All Vehicles		Period from 01 April to 31 March			
	Parking is not permitted on access road or verges			90p	£1.60	£3.00	£4.50	N/A

ТҮРЕ	CAR PARKS IN WHICH VALID	PERIOD OF WAITING	CHARGE
ANNUAL (Non-Residents and Business Owners as defined in Off Street Parking Places Order)	All Car Parks	Unlimited except as specified below: Saxon Square, Bank Close, Bridge Street and Wick Lane – up to 2 hours By Pass short stay area – up to 2 hours Priory – up to 4 hours (1 April – 30 September only) Highcliffe Castle – up to 4 hours NO RETURN WITHIN THREE HOURS	£370.00 - 01 April to 31 March £240.00 - 01 April to 30 September £215.00 - 01 October to 31 March
ANNUAL (Residents)	All Car Parks	Unlimited except as specified below: Saxon Square, Bank Close, Bridge Street and Wick Lane – up to 2 hours By Pass short stay area – up to 2 hours Priory – up to 4 hours (1 April – 30 September only) Highcliffe Castle – up to 4 hours NO RETURN WITHIN THREE HOURS	£225.00 - 01 April to 31 March £133.00 - 01 April to 30 September £118.00 - 01 October to 31 March
SHOPPING	Saxon Square, By-Pass, Pit Site, Law Courts, Wick Lane, Bridge Street, Bank Close, Wortley Road, Willow Way, Leisure Centre and Purewell Dairy	All car parks except Leisure Centre - up to 2 hours Leisure Centre - up to 3 hours NO RETURN WITHIN THREE HOURS	£ 77.00 - 01 April to 31 March £ 46.00 - 01 October to 31 March £ 50.00 - 01 September to 31 March
LEISURE CENTRE	Leisure Centre only	Once per day - up to 4 hours	£ 47.00 - 01 April to 31 March £ 26.00 - 01 October to 31 March
BOWLS CLUB	Leisure Centre only	Once per day - up to 4 hours weekdays - up to 5 hours weekends	£ 32.00 - 01 October to 30 April

F

ТҮРЕ			
	CAR PARKS IN WHICH VALID	PERIOD OF WAITING	CHARGE
HEAVY COMMERCIAL VEHICLES (Non-residents and Business Owners as defined in Off Street Parking Places Order)	Pit Site, Highcliffe Top	Unlimited	£1,110.00 - 01 April to 31 March £720.00 - 01 April to 30 September £645.00 - 01 October to 31 March
HEAVY COMMERCIAL VEHICLES (Residents)	Pit Site, Highcliffe Top	Unlimited	£675.00 - 01 April to 31 March £399.00 - 01 April to 30 September £354.00 - 01 October to 31 March
GOODS VEHICLES (Non-residents and Business Owners as defined in Off Street Parking Places Order)	All Car Parks except Highcliffe Castle, Christchurch Quay, Priory, Southcliffe Road, Steamer Point	All car parks except Saxon Square, Bypass, Bank Close, Bridge Street – Unlimited Saxon Sq, Bypass, Bank Close, Bridge St – 2 hours NO RETURN WITHIN THREE HOURS	£1,110.00 - 01 April to 31 March £720.00 - 01 April to 30 September £645.00 - 01 October to 31 March
GOODS VEHICLES (Residents)	All Car Parks except Highcliffe Castle, Christchurch Quay, Priory, Southcliffe Road, Steamer Point	All car parks except Saxon Square, Bypass, Bank Close, Bridge Street – Unlimited Saxon Sq, Bypass, Bank Close, Bridge St – 2 hours NO RETURN WITHIN THREE HOURS	£675.00 - 01 April to 31 March £399.00 - 01 April to 30 September £343.00 - 01 October to 31 March
		DISCOUNTS None	
	REFUNDS 70% of	value outstanding for remaining complete calendar me	onths
	REPLA	CEMENTS ALL CATEGORIES £10.00	
ANNEXURE

ON-STREET PLANS

Area Code	0	1	2	3	4	5	6	Page No.
SZ0997NEN	01.03.03							143
SZ0997NES	01.03.03							144
SZ0997SEN	01.03.03							145
SZ1097NWN	01.03.03							146
SZ1097NWS	01.03.03							140
								147
SZ1097NES	01.03.03							
SZ1097SWN	01.03.03							149
SZ1097SEN	01.03.03							150
SZ1197SWN	01.03.03							151
SZ1197SEN	01.03.03							152
SZ1197SES	01.03.03							153
SZ1297SES	01.06.07							154
SZ1393NEN	01.03.03							155
SZ1393NES	01.03.03							156
SZ1394NEN	01.03.03							157
SZ1394NES	01.03.03							158
SZ1394SEN	01.03.03							159
SZ1394SES	01.03.03							160
SZ1395SES	01.03.03			1				161
SZ1492NEN	01.03.03							162
SZ1492NES	01.03.03							163
SZ1492NES		01.06.07						164
	01.03.03	01.06.07				1	1	
SZ1493NWN	01.03.03	04.55.5	A A A A A A A A A A	0 4 5 - 5 - 5				165
SZ1493NWS	01.03.03	01.03.04	01.04.05	01.07.06				166
SZ1493NEN	01.03.03	01.04.05						167
SZ1493NES	01.03.03							168
SZ1493SWN	01.03.03	01.07.06						169
SZ1493SEN	01.03.03			1		1	1	170
SZ1493SES	01.03.03							171
SZ1494NWN	01.03.03							172
		01.06.07						
SZ1494NWS	01.03.03	01.06.07						173
SZ1494SWN	01.03.03							174
SZ1494SWS	01.03.03							175
SZ1494SEN	01.03.03							176
SZ1494SES	01.03.03							177
SZ1495SWN	01.03.03							178
SZ1495SWS	01.03.03							179
SZ1592NWN	01.03.03							180
		04.04.05	04.07.00					
SZ1592NWS	01.03.03	01.04.05	01.07.06					181
SZ1592NEN	01.03.03	01.03.04	01.07.06	01.06.07				182
SZ1592NES	01.03.03	01.03.04	01.04.05	01.07.06	01.06.07			183
SZ1592SWN	01.03.03							184
SZ1592SEN	01.03.03	01.07.06						185
SZ1593NWN	01.03.03	01.07.06						186
SZ1593NWS	01.03.03	01.04.05						187
SZ1593SWN	01.03.03	01.07.06	01.06.07					188
SZ1593SWS			01.00.07					189
	01.03.03	01.06.07						
SZ1593SEN	01.03.03	01.04.05						190
SZ1593SES	01.03.03	01.04.05	01.07.06					191
SZ1594SWS	01.03.03							192
SZ1595SEN	01.03.03							193
SZ1692NWN	01.03.03	01.06.07						194
SZ1692NWS	01.03.03					1	1	195
SZ1692NEN	01.03.03	01.06.07	01.06.07			1	1	196
SZ1692NES	01.03.03	000.07	000.07					190
SZ1692SWN	01.03.03							197
						-	-	
SZ1693NWS	01.03.03							199
SZ1693NES	01.03.03							200
SZ1693SWN	01.03.03	01.07.06						201
SZ1693SWS	01.03.03	01.03.04	01.07.06					202
SZ1693SEN	01.03.03							203
SZ1693SES	01.03.03	01.07.06						204
SZ1694NWN	01.03.03							205
SZ1694NEN	01.03.03	01.07.06						205
		01.07.00						
SZ1694SWN	01.03.03							207
SZ1694SWS	01.03.03							208
SZ1694SEN	01.03.03							209
SZ1695NWN	01.03.03							210
SZ1695NWS	01.03.03	01.06.07						211
SZ1695SWN	01.03.03	01.06.07		1		1	1	212
SZ1695SWS	01.03.03	01.06.07				1		213
SZ1695SES	01.03.03	01.00.07						213
			<u> </u>					
	01.03.03							215
SZ1696SWS SZ1792NWN	01.03.03	01.04.05	01.07.06					216

Area Code	0	1	2	3	4	5	6	Page No
SZ1792NWS	01.03.03	01.04.05	01.07.06					217
SZ1792NEN	01.03.03							218
SZ1792NES	01.03.03	01.07.06						219
SZ1792SWN	01.03.03							220
SZ1792SWS	01.03.03	01.04.05						221
SZ1792SEN	01.03.03	01.04.05	01.07.06	01.06.07				222
SZ1792SES	01.03.03	01.04.05	01.07.06					223
SZ1793NWN	01.03.03	01101100	0.1101100					224
SZ1793NWS	01.03.03	01.06.07						225
SZ1793NEN	01.03.03	01.06.07						225
		01.00.07						
SZ1793NES	01.03.03							227
SZ1793SWN	01.07.06							228
SZ1793SWS	01.03.03	01.07.06	01.06.07					229
SZ1793SEN	01.03.03	01.07.06						230
SZ1793SES	01.03.03							231
SZ1891NWN	01.03.03							232
SZ1892NWN	01.03.03							233
SZ1892NWS	01.03.03							234
SZ1892NEN	01.03.03							235
SZ1892NES	01.03.03							236
SZ1892SWN	01.03.03	01.04.05						237
SZ1892SWS	01.03.03	01.04.05	01.07.06					238
SZ1892SEN		01.04.05	01.07.00					230
	01.03.03	01.04.05						
Z1892SES	01.03.03	04.00.07						240
SZ1893NWN	01.03.03	01.06.07						241
SZ1893NWS	01.03.03	01.06.07						242
SZ1893NEN	01.03.03							243
SZ1893NES	01.03.03							244
SZ1893SWN	01.03.03	01.04.05						245
SZ1893SWS	01.03.03							246
SZ1893SEN	01.03.03	1						247
SZ1893SES	01.03.03	01.04.05	01.07.06					248
SZ1894SEN	01.03.03	01.01.00	01.01.00					249
SZ1894SES	01.03.03							250
SZ1992NWN	01.03.03							250
		01.04.05						251
SZ1992NWS	01.03.03	01.04.05						
SZ1992NEN	01.03.03							253
SZ1992SWN	01.03.03							254
SZ1993NWS	01.03.03							255
SZ1993NEN	01.03.03							256
SZ1993NES	01.03.03							257
SZ1993SWN	01.03.03							258
SZ1993SWS	01.03.03							259
SZ1993SEN	01.03.03							260
SZ1994NWS	01.03.03							261
SZ1994NES	01.03.03							262
SZ1994SWN	01.03.03							263
SZ1994SEN	01.03.03							203
SZ1994SES	01.07.06							265
SZ2093NWN	01.03.03							266
SZ2093NWS	01.03.03							267
SZ2093NES	01.03.03							268
SZ2093SWN	01.03.03	01.06.07						269
SZ2093SWS	01.03.03							270
SZ2093SEN	01.03.03							271
SZ2093SES	01.03.03							272
SZ2094SWN	01.03.03	1						273
SZ2094SWS	01.03.03							274
SZ2094SES	01.04.05	01.07.06						275
SZ20940E0	01.03.03	01.03.04						276
SZ2193NWS	01.03.03	01.03.04						277
Z2193NEN	01.03.03	01.04.05	01.07.00					278
Z2193NES	01.03.03	01.04.05	01.07.06					279
Z2193SWN	01.03.03							280
Z2193SWS	01.03.03							281
Z2193SEN	01.03.03							282
SZ2194NWN	01.03.03							283
SZ2194NWS	01.03.03	01.07.06	01.06.07					284
SZ2194NES	01.03.03	01.06.07						285
SZ2194SWN	01.03.03							286
SZ2194SWS	01.04.05	01.07.06						287
SZ2194SEN	01.04.05	01.01.00						287
SZ2293NWS	01.07.06							289
SZ2294NWS	01.03.03							290

Page 141

ACTIVE DATES OF EACH PLAN SHEET **REVISION NUMBER**



Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON		Christchurch Borough Council, Civic Offices, Bridge Street,
Sheet Active From Date 01.03.03	AND OFF-STREET PARKING PLACES) ORDER 2007		Christchurch, Dorset BH23 1AZ Head of Planning & Transportat
Scale 1:1250	Area Code SZ0997NES		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\	Prints - 20070401\SZ0997-REV0-20070401.dwg	Page 144	The Ordnance Survey information on the of the Controller of Her Majesty's Statio Unauthorized reproduction infringes Croproceedings.













ANE No waiting at any time			
Golf Course Twr			
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES)		Christchurch Borough Council,
Sheet Active From Date 01.03.03	(CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007		Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportat
Scale 1:1250	Area Code SZ1097SEN		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace	\Prints - 20070401\SZ1097-REV0-20070401.dwg	Page 150	The Ordnance Survey information on the of the Controller of Her Majesty's Station Unauthorized reproduction infringes Croproceedings.















			I O BURNER Resolution Resolu
		Mar War	
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON		Christchurch Borough Council, Civic Offices, Bridge Street,
Sheet Active From Date 01.03.03	AND OFF-STREET PARKING PLACES) ORDER 2007		Christchurch, Dorset BH23 1AZ
Scale 1:1250	Area Code SZ1394NEN		Head of Planning & Transportatio David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\F	Prints - 20070401\SZ1394-REV0-20070401.dwg	Page 157	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.



nis drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil







Golf Course		CUNIFER CLOSE	ses // //
Post	ost		BARN CL.
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES)		Christchurch Borough Council, Civic Offices, Bridge Street,
Sheet Active From Date 01.03.03	(CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	$ $ \bigwedge	Christchurch, Dorset BH23 1AZ Head of Planning & Transportation
Scale 1:1250	Area Code SZ1395SES		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\P	Prints - 20070401\SZ1395-REV0-20070401.dwg	Page 161	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.



his drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil



























Image: second			
Sheet Revision Number O Sheet Active From Date 01.03.03	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	N	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ
^{Scale} 1:1250	Area Code SZ1494SWS		Head of Planning & Transportatio David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\F	Prints - 20070401\SZ1494-REV0-20070401.dwg	Page 175	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.



nis drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil


		Poly Tier +	FAIRMILE
			Age Age No stopping
			except buses 7am - 7pm
			The Howard Social Centre
Playground			
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON	λ I	Christchurch Borough Council, Civic Offices, Bridge Street,
Sheet Active From Date 01.03.03	AND OFF-STREET PARKING PLACES) ORDER 2007		Christchurch, Dorset BH23 1AZ Head of Planning & Transportatio
^{Scale} 1:1250	Area Code SZ1494SES		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\	Prints - 20070401\SZ1494-REV0-20070701.dwg	Page 177	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.



nis drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil



HILISIDE DRIVE			
Image: No Stopping			attillite attillite attillite Tumulus
except buses 7am - 7pm e e e s s			Reservoir (disused)
		G _e	Are Ba
1 1 1 <th></th> <th></th> <th>es la la</th>			es la
0	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON		Christchurch Borough Council, Civic Offices, Bridge Street,
Sheet Active From Date 01.03.03	AND OFF-STREET PARKING PLACES) ORDER 2007		Christchurch, Dorset BH23 1AZ Head of Planning & Transportatio
Scale 1:1250	Area Code SZ1495SWS		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\F	Prints - 20070401\SZ1495-REV0-20070401.dwg	Page 179	The Ordnance Survey information on this of the Controller of Her Majesty's Stationer Unauthorized reproduction infringes Crown proceedings.









































			A A No stopping except buses 7an - 7pm
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES)		Christchurch Borough Council,
Sheet Active From Date 01.03.03	(CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	N	Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportat
^{Scale} 1:1250	Area Code SZ1693NWS		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\works	pace\Prints - 20070401\SZ1693-REV2-20070401.dwg	Page 199	The Ordnance Survey information on th of the Controller of Her Majesty's Statio Unauthorized reproduction infringes Cro proceedings.



nis drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil















	excep	topping touses 7pm		
Sheet Revision Number O Sheet Active From Date 01.03.0)3	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	N	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transporta
Scale 1:1250		Area Code SZ1694SWN		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I. The Ordnance Survey information on th
H:\LTS\CAD\Dr	awings\CadC\workspace\I	Prints - 20070401\SZ1694-REV1-20070401.dwg	Page 207	of the Controller of Her Majesty's Sta Unauthorized reproduction infringes proceedings.



this drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. rown Copyright and may lead to prosecution or civil






















this drawing has been reproduced with the kind permission ionery Office © Crown Copyright Licence No.100024379. Frown Copyright and may lead to prosecution or civil



























	1 1		
Mud MHW Mud summer MNN Shingle		Landing Stage	Marsh Marsh
Mud			
Christchurch Hark Mud	pour	Mud	
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH		Christehurch Borough Council
Sheet Active From Date 01.03.03	(VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	N	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportati
Scale 1:1250	Area Code SZ1891NWN		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace	\Prints - 20070401\SZ1891-REV0-20070401.dwg	Page 232	The Ordnance Survey information on thi of the Controller of Her Majesty's Station Unauthorized reproduction infringes Cro proceedings.



nis drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil

No waiting Mon-Fri Ban - 6pn 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4 4	r Beaver Industrial Estate		
Sheet Revision Number O Sheet Active From Date 01.03.03	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES)	N	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ
UT.UJ.UJ Scale	ORDER 2007		Head of Planning & Transportatio David Jobbins BSc (Hons)
1:1250	SZ1892NWN		DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\Prints - 20070401\SZ1892-REV2-20070401.dwg		Page 233	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.



his drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. Town Copyright and may lead to prosecution or civil






























			MP 102
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES)		Christchurch Borough Council,
Sheet Active From Date 01.03.03	(CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	\square	Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportation
Scale 1:1250	Area Code SZ1894SEN		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace	Prints - 20070401\SZ1894-REV0-20070401.dwg	Page 249	The Ordnance Survey information on this of the Controller of Her Majesty's Station Unauthorized reproduction infringes Crow proceedings.



is drawing has been reproduced with the kind permission nery Office © Crown Copyright Licence No.100024379. wwn Copyright and may lead to prosecution or civil











No stopping except buses 7am - 7pm		Hoburne Caravan Park	El Lub Sta
			Bu
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES)		Christchurch Borough Council,
Sheet Active From Date 01.03.03	(CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	$ $ \bigwedge	Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportati
^{Scale} 1:1250	Area Code SZ1993NWS		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace	\Prints - 20070401\SZ1993-REV0-20070401.dwg	Page 255	The Ordnance Survey information on thi of the Controller of Her Majesty's Station Unauthorized reproduction infringes Cro proceedings.



nis drawing has been reproduced with the kind permission onery Office © Crown Copyright Licence No.100024379. own Copyright and may lead to prosecution or civil





































QQ QQ 00 00 00 00 00 00 00 00 00 00 00 0			
		Pissues	
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH		
Sheet Active From Date 01.03.03	(VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007	Ν	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ
Scale	Area Code		Head of Planning & Transportation David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
1:1250 Cad Ref	SZ2094SWS		The Ordnance Survey information on this of
H:\LTS\CAD\Drawings\CadC\workspace	\Prints - 20070401\SZ2094-REV2-20070401.dwg	Page 274	of the Controller of Her Majesty's Stationer Unauthorized reproduction infringes Crowr proceedings.



is drawing has been reproduced with the kind permission nery Office © Crown Copyright Licence No.100024379. wn Copyright and may lead to prosecution or civil






























		BM 40.25m	
El Sub Sta			58 60 64 66 EL Sub S
20 20 20 20 20 20 20 20 20 20 20 20 20 2	et Gardens		
BREIDATL-AMUSS CLUBSE 6 6 7 8 1 8 1 8 1 1 8 1 1 8 1 1 8 1 1 8 1			
No stopping except buses 7am - 7pm	TE LA		A CHEWTON FARM
Sheet Revision Number	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON		Christchurch Borough Council, Civic Offices, Bridge Street,
Sheet Active From Date 01.03.03	AND OFF-STREET PARKING PLACES) ORDER 2007		Christchurch, Dorset BH23 1AZ Head of Planning & Transportat
Scale 1:1250	Area Code SZ2294NWS		David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\	Prints - 20070401\SZ2294-REV0-20070401.dwg	Page 290	The Ordnance Survey information on th of the Controller of Her Majesty's Station Unauthorized reproduction infringes Cro proceedings.



ANNEXURE

OFF-STREET PLANS

	0	1	2	3	4	5	6	7	8	9	10	Page No
AVON RUN No 1	27.01.03	01.03.04	01.06.06									293
BANK CLOSE No 2	27.02.02	01.03.04	01.04.05	01.06.06								294
BRIDGE STREET No 3	27.01.03	01.03.04	01.06.07									295
BYPASS No 4	27.01.03	01.04.05	01.06.06									296
CHRISTCHURCH QUAY No 5	27.02.02	01.03.04	01.04.05	01.06.07								297
HIGHCLIFFE CASTLE NORTH No 6	05.03.02	01.06.06										298
HIGHCLIFFE CASTLE SOUTH No 6	05.03.02	01.03.04	01.06.06	01.06.07								299
HIGHCLIFFE CLIFF TOP No 7	27.02.02	01.06.06	01.06.07									300
HIGHCLIFFE CLIFF TOP GRASS AREA No 7	27.02.02	01.06.07										301
LAW COURTS No 8	27.01.03	01.03.04	01.06.07									302
LEISURE CENTRE EAST No 9	27.01.03	01.03.04	01.06.06	01.06.07								303
LEISURE CENTRE WEST No 9	27.01.03	01.03.04	01.06.06	01.06.07								304
MAYORS MEAD No 10	27.01.03	01.03.04	01.06.06	01.06.07								305
MUDEFORD QUAY No 11	27.01.03	01.03.04	01.06.06									306
PIT SITE No 12	05.03.02	01.03.04	01.04.05	01.06.06	01.06.07							307
PRIORY CHURCH No 13	27.01.03	01.03.04	01.06.06									308
PUREWELL DAIRY No 14	27.02.02	01.03.04	01.06.06	01.06.07								309
SAXON SQUARE (Lower) No 15	27.01.03											310
SAXON SQUARE (Upper) No 15	27.01.03	01.03.04	01.04.05	01.06.06	01.06.07							311
SOUTHCLIFFE ROAD No 16	27.02.02	01.04.05	01.06.06									312
STEAMER POINT No 17	27.01.03	01.03.04	01.04.05	01.06.07								313
WICK LANE No 18	27.01.03	01.03.04	01.06.06	01.06.07								314
WILLOW WAY No 19	27.02.02	01.03.04	01.06.06	01.06.07								315
WORTLEY ROAD No 20	27.02.02	01.03.04	01.06.06									316
CAR PARK SYMBOLS	01.03.03	01.03.04	01.04.05	01.06.06								317
												_

Page 292

ACTIVE DATES OF EACH OFF-STREET PLAN SHEET REVISION NUMBER

THIS SHEET REVISION NUMBER 3 THIS SHEET ACTIVE FROM 01.06.07

		AVON RUN ROAD	
10 5 0 1	0 20 30 metres)		
Spaces 224 Plus Disabled 3 Recycling 0 On Grass/ Gravel n\a	Car Park AVON RUN NO. 1 Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\carparks\prints\ \Avon Run\PRINT.REV2.20070601.dwg	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007 Sheet Revision Number 2 Sheet Active From Date 01.06.06 Scale As above Page 293	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportat David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I. The Ordnance Survey information on th of the Controller of Her Majesty's Station Unauthorized reproduction infringes Cro proceedings.



























The Ordnance Survey information on this drawing has been reproduced with the kind permission of the Controller of Her Majesty's Stationery Office © Crown Copyright Licence No.100024379. Unauthorized reproduction infringes Crown Copyright and may lead to prosecution or civil





	Access to rear of Roosters Public House R and R a	Recycling Area	Conservative Club Private Parking Vehicular access to Conservative Club cellar KEEP CLEAR FR BOD FR CONSERVATIVE Club Cellar CONSERVATIVE Club Cellar CONSERVATIVE Club Cellar CONSERVATIVE Club Cellar FR CONSERVATIVE CELLAR FR CONSERVATI
Spaces 57	Car Park	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES)	
Plus		(CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES)	Christchurch Borough Council, Civic Offices, Bridge Street,
Disabled 2	PIT SITE	ORDER 2007 Sheet Revision Number	Christchurch, Dorset BH23 1AZ
HGV 2	No. 12	Sheet Active From Date 01.06.07	Head of Planning & Transportatio David Jobbins BSc (Hons)
Open Area c.9		Scale As above	DipT.P., DipE.M., M.R.T.P.I.
' On Grass/ Gravel n∖a	Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\carparks\prints\ \Pit Site\PRINT.REV3.20070601.dwg	Page 307	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.









MIL Stream
ion Christchurch where time is pleasant

Unauthorized reproduction infringes Crown Copyright and may lead to prosecution or civil





The Ordnance Survey information on this drawing has been reproduced with the kind permission of the Controller of Her Majesty's Stationery Office © Crown Copyright Licence No.100024379. Unauthorized reproduction infringes Crown Copyright and may lead to prosecution or civil





The Ordnance Survey information on this drawing has been reproduced with the kind permission of the Controller of Her Majesty's Stationery Office © Crown Copyright Licence No.100024379. Unauthorized reproduction infringes Crown Copyright and may lead to prosecution or civil









Have you purchased and displayed	1	Trailer carpark with time limit	32	Access Way Do Not Obstruct	67
Pay and display carpark	٢	Mooring and dinghy park	<u>3</u> 3	Duck Crossing	68
Disabled parking	3	No right turn	34)	No Unauthorised Parking	69
No entry	<u>(4)</u>	Coaches not allowed	<u>40</u>	Information Sign	\bigcirc
No exit	5	Parking conditions for permit holders	(41)	Short Stay Parking Sign	71
To shops, tourist information		Reserved bays with time limit	42 43	No Pedestrians Signs	12
	6	Reserved bays	(43)	Bin (circular)	<u> </u>
Direction indicator	\bigcirc	Slow pedestrians	(44)	Bin (circular concrete)	Ø
Heavy vehicles, large vans park here	8	Bicycle parking	(45)	Bin (rectangular conçrete)	A
Give way	9	Direction indicator to coaches	46	Bin (normal suspended)	B
Overflow carpark indication	10	Parking conditions for trailers	() ()	Lamp post (2 lights)	\bigcirc
Litter act	Ĩ			Lamp post (1 light)	œ
Toilets	12	Slipway charging	48	Lamp post (1 light)	×
Maximum height	13	Public slipway	49	Stairs/steps	Ô
Exit		No ball games allowed	50	Ramp	D
	14	Disabled parking wheelchair users only	51	Ramp (pedestrians)	Ē
Chevron	15	No parking or turning facilities beyond this point	52	Drain covers	
Speed limit	16	Private entrance	<u>6</u> 3	Gullys	圓
Valid ticket must be displayed	(17)	Caution traffic from right	<u>5</u> 4	Manholes	\mathbb{U}
No parking on verges	18	Beware carpark traffic		Footpath	F
Pedestrian area	(19)		<u>65</u>	Electrics box	
No parking beyond this point	20	Children to be accompanied by an adult	56	Hut	H
Pay and display beyond this point	21	Area and safety sign	67	Doggy box	
Returh trolleys here	22)	No parking lorries turning	58	Concrete post Timber post	×
Public slipway	23)	Disabled parking only beyond this point	59	Emergency gate/access	Ø
Pay and display (vehicle excise)	24) 25	No water scooters	60		G
Warning of ramps	25	Loading Area Permitted Vehicles Only	61	Gate/barrier	Ĥ
Automatic barrier pedestrians KEEP CLEAR	26)	No Waiting At AnyTime	62)	Vehicle access	K
This area forms part of the pay and display carpark/ A valid ticket must be purchased	27	CCTV Sign	63	Tree	O
No vehicles beyond this point	28	Motorcycles Prohibited	64	Pedestrian access	(L)
No parking	29	No Skateboarding No Rollerblades	65	Machine with pay here and general sign	[11]°
Directions to heavy vehicles and large vans	30	Gates Closed Daily Apr-Sept 2100-900	66	Recycling Skip	M
Pay and display trailer carpark			~	CCTV Camera Column	N
	31				

CAR PARK SYMBOLS	THE BOROUGH OF CHRISTCHURCH (VARIOUS ROADS AND PARKING PLACES) (CONSOLIDATION OF ORDERS MADE FOR ON AND OFF-STREET PARKING PLACES) ORDER 2007 Sheet Revision Number 3 Sheet Active From Date 01.06.06 Scale n/a	Christchurch Borough Council, Civic Offices, Bridge Street, Christchurch, Dorset BH23 1AZ Head of Planning & Transportatic David Jobbins BSc (Hons) DipT.P., DipE.M., M.R.T.P.I.
Cad Ref H:\LTS\CAD\Drawings\CadC\workspace\carparks\prints\ \SYMBOLS.PRINT\SYMBOLS.PRINT.REV3.20070601.dwg	Page 317	The Ordnance Survey information on this of the Controller of Her Majesty's Statione Unauthorized reproduction infringes Crow proceedings.



s drawing has been reproduced with the kind permission hery Office © Crown Copyright Licence No.100024379. wn Copyright and may lead to prosecution or civil